Quantitative examination of lithic raw material variability and fracture predictability in the Middle and Upper Paleolithic assemblages in southern Jordan

(南ヨルダンの中部・上部旧石器石器群における石器石材の多様性と 剥離予測性に関する定量的研究)

> SUGA, Eiki (須賀 永帰)

A dissertation for the degree of Doctor of Science Department of Earth and Environmental Sciences, Graduate School of Environmental Studies, Nagoya University (名古屋大学大学院環境学研究科地球環境科学専攻学位論文 博士(理学))

2024

Contents

Chapter 1. Introduction1
1.1. Middle-to-Upper Paleolithic transition1
1.2. Brief overview of lithic raw material studies in the Levantine Paleolithic3
1.3. Aims and organization of the research5
Chapter 2. Research materials
2.1. Lithic assemblages from the Jebel Qalkha area, Jordan9
2.2. Geological settings in the Western Hisma Basin12
Chapter 3. Diachronic changes in lithic raw material (chert) use
3.1. Methods
3.2. Results
3.3. Discussion25
3.4. Summary
Chapter 4. Quantitative examination of chert mechanical properties and fracture
Chapter 4. Quantitative examination of chert mechanical properties and fracture predictability
Chapter 4. Quantitative examination of chert mechanical properties and fracture predictability
Chapter 4. Quantitative examination of chert mechanical properties and fracture predictability
Chapter 4. Quantitative examination of chert mechanical properties and fracture predictability 36 4.1. Background of fracture predictability 36 4.2. Methods 40 4.3. Results 43
Chapter 4. Quantitative examination of chert mechanical properties and fracture predictability 36 4.1. Background of fracture predictability
Chapter 4. Quantitative examination of chert mechanical properties and fracture predictability 36 4.1. Background of fracture predictability
Chapter 4. Quantitative examination of chert mechanical properties and fracture predictability 36 4.1. Background of fracture predictability
Chapter 4. Quantitative examination of chert mechanical properties and fracture predictability 36 4.1. Background of fracture predictability. 36 4.2. Methods 40 4.3. Results 43 4.4. Discussion. 45 4.5. Summary. 49 Chapter 5. Discussion 56 5.1. Quantitative characterization of lithic raw material variations and fracture 56
Chapter 4. Quantitative examination of chert mechanical properties and fracture predictability 36 4.1. Background of fracture predictability. 36 4.2. Methods 40 4.3. Results 43 4.4. Discussion 45 4.5. Summary. 49 Chapter 5. Discussion 56 5.1. Quantitative characterization of lithic raw material variations and fracture 56 5.2. Gluenee is the predictability 56
Chapter 4. Quantitative examination of chert mechanical properties and fracture predictability 36 4.1. Background of fracture predictability. 36 4.2. Methods 40 4.3. Results 43 4.4. Discussion 45 4.5. Summary. 49 Chapter 5. Discussion 56 5.1. Quantitative characterization of lithic raw material variations and fracture predictability. 56 5.2. Changes in the raw material selection at the MP-UP transition 58
Chapter 4. Quantitative examination of chert mechanical properties and fracture predictability 36 4.1. Background of fracture predictability 36 4.2. Methods 40 4.3. Results 43 4.4. Discussion 45 4.5. Summary 49 Chapter 5. Discussion 56 5.1. Quantitative characterization of lithic raw material variations and fracture predictability 56 5.2. Changes in the raw material selection at the MP-UP transition 58 5.3. Further research issues 60

Acknowledgements	
References	66
Appendix	90

Chapter 1. Introduction

1.1. Middle-to-Upper Paleolithic transition

The transition from the Middle Paleolithic (MP) to the Upper Paleolithic (UP) has long been an important issue in prehistoric archaeology (e.g., Garrod, 1951, 1955) and also gained greater attention recently. The Middle to Upper Paleolithic transition (hereafter the MP-UP transition) is known for its temporal proximity to key paleoanthropological processes, such as the wide geographic dispersal of Homo sapiens (ca. 50-40 ka) and their interaction and interbreeding with Neanderthals and Denisovans (Shea, 2013; Higham et al., 2014; Bosch et al., 2015; Bae et al., 2017; Dennell, 2020; Kuzmin et al., 2022; Vallini et al., 2022; Slimak, 2023). The MP-UP transition has been considered as an important topic in various fields. One of the main research issues is how paleoanthropological processes related to archaeological records. were Paleoanthropological studies (e.g., ancient DNA) alone cannot explain the factors and complicated processes of the major events in human history mentioned above, thus it is necessary to examine the archaeological record. Especially, some studies focused on cultural remains in the Levantine region, which was the beginning of Homo sapiens geographic dispersal across Eurasia (For the Levantine case studies, see Akazawa et al., 1998; Bar-Yosef, 2000; Henry, 2003; Shea, 2008; Meignen, 2012; Douka, 2013; Kadowaki, 2013; Rose and Marks, 2014; Belfer-Cohen and Goring-Morris, 2018; Abadi et al., 2020; Dennell, 2020; Stutz, 2020; Boaretto et al., 2021). Consequently, the MP-UP transition in the Levant is known for changes in a wide range of archaeological records, such as lithic technology, foraging practices, mobility patterns, and personal adornments (Marks, 1981, 1983b; Rabinovich, 2003; Kuhn et al., 2009; Belfer-Cohen and Hovers, 2010; Shea and Sisk, 2010; Douka, 2013; Stiner et al., 2013; Henry et al., 2017; Goder-Goldberger and Malinsky-Buller, 2022).

Several studies employed a taxonomic term of the "Initial Upper Paleolithic (IUP)" as a chronological framework in understanding these transitional archaeological records distributed over a wide geographical region (Kuhn and Zwyns, 2014; Kuhn, 2019; Goring-Morris and Belfer-Cohen, 2020; Hublin et al., 2020; Zwyns, 2021; Carvalho and Bicho, 2022; Kot et al., 2022). Moreover, the "Late Middle Paleolithic (LMP)" is chronologically distinguished before the IUP, and the "Early Upper Paleolithic (EUP)" is chronologically distinguished after the IUP. These chronological entities were defined using techno-typological criteria of many lithic assemblages in the Levant (Fig.1.1.; Garrod, 1951; Copeland, 1975, 2000; Marks and Volkman, 1983; Gilead, 1991; Richter et al., 2001; Pastoors et al., 2008; Meignen, 2012; Shea, 2013; Rose and Marks, 2014; Kadowaki et al., 2019a, 2021; Goder-Goldberger, 2020; Goder-Goldberger and Mailnsky-Buller, 2022). More specifically, in the LMP (75–50 ka), the Levallois method is dominantly employed. Among the Levallois flaking modes, the unidirectional convergent mode is dominant, accompanied by the bidirectional and centripetal modes. Non-Levallois reduction methods include single platform cores, bladelet cores and cores on flakes. The retouched tool categories are characterized by abundant side-scrapers (Nishiaki et al., 2012; Malinsky-Buller et al., 2014; Sharon and Oron, 2014; Baykara et al., 2015; Goder-Goldberger and Bar-Matthews, 2019; Goder-Goldberger et al., 2020). In the IUP (50-35 ka), the lithic technology shifted to increasing occurrences of robust blades knapped with hard hammers. The retouched tool types are characterized by UP tool types, such as end-scrapers and burins. Elongated points, including typological Levallois points, are also found (Kuhn et al., 2009; Leder, 2016; Nishiaki, 2018; Kadowaki et al., 2019b, 2022a; Barzilai, 2022; Goder-Goldberger et al., 2023). Moreover, some studies proposed several distinct variants of IUP lithic industries based on the appearance of unique tool types, such as Emireh points and chamfered pieces (Goring-Morris and Belfer-Cohen, 2020). In the EUP (45-30 ka), bladelets increased, and their small striking platforms with fine overhang removals indicate the employment of softhammer percussion. Within the period of the EUP, many archaeologists recognize the subdivision of two techno-complexes, Ahmarian and Aurignacian. The Ahmarian, examined in this dissertation, is characterized by the composition of unique tool types, such as backed points and el-Wad points (Bar-Yosef and Befer-Cohen, 2019; Shea et al., 2019; Parow-Souchon et al., 2021; Abulafia et al., 2021; Shemer et al., 2023).

In contrast to these numerous studies on lithic techno-typology, little is known about whether any changes occurred in the use of lithic raw materials at the MP-UP transition in the Levant. At this stage, only a few studies have analyzed lithic raw material changes at the Levantine MP-UP transition in contrast to the other regions like Europe (For the European case studies, see Féblot-Augustins, 2009; Grimaldi et al., 2014; Aubry et al., 2016; Tomasso and Porraz, 2016; Negrino and Riel-Salvatore, 2018; Riel-Salvatore and Negrino, 2018; Holt et al., 2019). Therefore, this dissertation focuses on lithic raw material utilization at the MP-UP transition in the Levant. In the next section, I review the case studies of lithic raw material in the Levant from a methodological viewpoint and summarize some research problems.

1.2. Brief overview of lithic raw material studies in the Levantine Paleolithic

During the entire Levantine Paleolithic, chert (flint) is mainly used as raw material of flaked stone tools (Delage, 2003; Delage and Webb, 2020). Recent raw material studies actively adopted field surveys for chert outcrops and analyzed raw material samples collected at the outcrops in comparison to archaeological samples (Finkel et al., 2016, 2018, 2019, 2020b, 2022; Groucutt et al., 2017; Parow-Souchon et al., 2021). More specifically, Finkel et al. (2016, 2018, 2019, 2020b) aimed to identify extraction and reduction (E&R) complexes from the lithic assemblage composition collected from raw material sources. The field survey of raw material outcrops is often regarded as preliminary studies for identifying raw material sources.

Increasingly, the identification of raw material sources was conducted through various geochemical studies such as INAA (Julig et al., 2007), the measurement of ¹⁰Be (Boaretto et al., 2009), ICP-MS (Ekshtain et al., 2014, 2017, 2019; Ekshtain and Tryon, 2019; Agam, 2020; Bellar et al., 2020; Finkel et al., 2020a, 2023; Ekshtain and Zaidner, 2022; Bellar, 2023), and petrographic analysis (Agam et al., 2020) of chert samples from outcrops and archaeological sites. Because the source identification analysis result is easily linked to lithic raw material procurement and exploitation strategies, Minimal Distance Analysis was performed by using ArcGIS (Parow-Souchon and Purschwitz, 2020). Some studies mentioned the raw material selectivity through the classification of chert's macroscopic aspects (Wilson et al., 2016; Agam, 2020; Agam et al. 2022) and the analyses of the quality (Delage, 2007; Hovers, 2009; Shimelmitz et al., 2020; Assaf, 2021) and use-wear (Agam and Zupancich, 2020) of chert artifacts. As described above, the classification of lithic artifacts by macroscopic aspects is still a primary step in examining the use of chert. Because geochemical analyses of lithic raw material can only be applied to a limited number of samples. On the other hand, the blind test about the macroscopic classification of chert types was also performed (Agam and Wilson, 2018). A few studies identified the sources of Paleolithic obsidian artifacts (Frahm and Hauck, 2017; Frahm and Tryon, 2019). From a wider perspective, some studies discussed lithic provisioning strategies at the UP (Kuhn, 2004) and the MP (Henry, 2011; Varoner et al., 2022).

Explaining the studies related to the MP–UP transition, Julig et al. (2007), conducted instrumental neutron activation analyses (INAA) to examine the changes in chert sources at the MP-UP transition, using geological samples and the lithic assemblage of Jerf al-Ajla Cave, Syria. They found that while MP inhabitants used local Eocene raw material almost exclusively, the use of more distant Cretaceous raw material increased in the Upper and Epipaleolithic periods. In Israel, the studies of 'Ein Qashish (Ekshtain et al., 2014), Amud Cave (Ekshtain et al., 2017), Skhul Cave (Ekshtain and Tryon, 2019) and

Nesher Ramla (Ekshtain and Zaidner, 2022) have recently revealed the raw material exploitation patterns in the MP. They indicate that MP inhabitants at these sites exploited lithic raw materials not only from local sources but also from mid-range or non-local sources.

In this way, lithic raw material studies in the Levantine Paleolithic are progressing in various themes, and focusing on the identification of raw material sources. In fact, to demonstrate Paleolithic raw material utilization, identifying raw material sources and considering models of raw material procurement and exploitation have been conducted as effective approaches in various regions (e.g., Shackley, 2008; Spinapolice, 2012; Ekshtain et al., 2014; Garvey, 2015; Aubry et al., 2016; Frahm and Hauck, 2017; Brandl et al., 2018; Soto et al., 2018; Agam, 2020; Gómez de Soler et al., 2020b; Valde-Nowak and Ciésla, 2020; Ekshtain and Zaidner, 2022). On the other hand, few studies focused on the capacity and availability of lithic raw material in the Levantine Paleolithic.

1.3. Aims and organization of the research

The problems mentioned above can be summarized as follows. Firstly, few Levantine studies focus on lithic raw material at the MP-UP transition. As mentioned above, a number of paleoanthropological research suggested that several major events in human evolution had occurred at the MP-UP transition. One of the further interests in this research topic is how material culture contributed to these events. Lithic artifacts are representative of the archaeological discussion of material culture in the Paleolithic. Moreover, lithic raw materials have a great influence on the production and modification of stone tools (Andrefsky, 1994, 2009). Lithic raw material studies play an important role in exploring prehistoric behavioral strategies because understanding how prehistoric people used lithic raw material can help us to explore their technological organization linked to mobility and land use (e.g., Binford, 1979; Nelson, 1991; Marks et al., 1991; Kuhn, 1995). Lithic raw material studies can provide us with the comprehension of the resource utilization at the MP-UP transition when material culture shifted. However, the discussion like mobility and land use is by no means abundant because of few lithic raw material studies in the Levant.

Secondly, these lithic raw material studies are slightly inclined toward the source identification. I think that due to the geological conditions in the Levant, where chert is mainly available as lithic raw materials, the variety of lithic raw material excavated from the sites is not abundant accordingly. The exiguity of lithic raw material variety has resulted in few opportunities for the research themes, which may have led to the paucity of case studies and the bias in analysis methods. In fact, the study in other region has adopted more various themes. For example, there are analyses of lithic raw material variety (e.g., 2019), raw material quality examination through quantifying mechanical properties (e.g., Webb and Domanski, 2008; Moník and Hadraba, 2016; Namen et al., 2022) and raw material transport patterns using cortex ratios (e.g., Lin et al., 2010, 2015, 2016). These methods can be sufficiently applied to the Levantine cases even if chert is only used for lithic raw material.

Here, this dissertation proposes a novel framework of lithic raw material study combining the macroscopic classification with the physical properties, which have not been analyzed extensively in the Levant. More specifically, I present analyses of four lithic assemblages of the LMP, IUP, and EUP (Ahmarian) excavated from the Jebel Qalkha area in southern Jordan to examine whether any changes occurred in raw material utilization at the MP-UP transition. The Jebel Qalkha sites are located close to each other within an area of 6 km², sharing essentially the same availability and accessibility of lithic

raw materials. One significance of this dissertation is the diachronic examination of lithic raw material selection and procurement through three continuous phases over the MP-UP transition, a landmark period in human history.

To approach the above issues, this dissertation has performed two analyses. First of all, Chapter 2 introduces Paleolithic sites in the Jebel Qalkha area and geological settings around the study area. Chapter 3 classifies chert types by using several macroscopic and microscopic attributes. Chapter 3 also presents diachronic changes in chert-type frequencies from the LMP to the EUP and discusses the relation to techno-typological attributes. Chapter 3 was revised from the article published in Archaeological Research in Asia (Suga et al., 2022). Furthermore, Chapter 4 also examines how the differences in macroscopic appearances (chert types) that were conventionally used as criteria for the chert "quality", are related to some quantitative attributes in hardness involved in flaking lithics. Chapter 4 was also revised from the article published in Journal of Paleolithic Archaeology (Suga et al., 2023). Although the two publications include co-authors, the author of this dissertation (Suga, E.) had a major role in the studies and wrote the manuscripts. Based on Chapters 3 and 4 results, Chapter 5 presents the significance of this dissertation in the lithic raw material studies. Chapter 5 also shows further problems of this dissertation and presents some preliminary analyses and results to overcome these issues. Chapter 6 shows the conclusion of this dissertation.



Fig. 1.1. Map of the Levant region showing the LMP, IUP and EUP sites. A red circle shows study sites. Vector and raster map data are acquired from Natural Earth (https://www.naturalearthdata.com/). Site names, periods, and references are as follows.

1) Üçağızlı Cave (IUP-EUP): Kuhn et al., 2009. 2) Dederiyeh Cave (LMP): Nishiaki et al., 2012.

3) Wadi Kharar 16R (EUP): Kadowaki et al., 2015. 4) Hummal (LMP): Hauck, 2011.

5) Umm el-Tlel (IUP): Boëda and Bonilauri, 2006. 6) Abou Halka (IUP): Leder, 2016.

7) Ksar Akil (IUP): Bosch et al., 2015. 8) Antelias (EUP): Copeland, 2000.

9) Manot (EUP): Alex et al., 2017. 10) Nahal Mahanyeem Outlet (LMP): Kalbe et al., 2014.

11) Amud Cave (LMP): Ekshtain et al., 2017. 12) Tabun Cave (LMP): Ronen, 2017.

- 13) 'Ein Qashish (LMP): Malinsky-Buller et al., 2014.
- 14) Kebara Cave (LMP-EUP): Meignen, 2019. 15) Mughr el-Hamamah (EUP): Shea et al., 2019.
- 16) Far'ah II (LMP): Goder-Goldberger et al., 2020.
- 17) Tor Sadaf (IUP): Fox and Coinman, 2004. 18) EHLPP1 (EUP): Clark et al., 2017.
- 19) Al-Ansab 1 (EUP): Schyle and Richter (Eds.), 2015.
- 20) Al-Ansab 2 (IUP): Schyle and Richter (Eds.), 2015.
- 21) Boker Tachtit (IUP): Boaretto et al., 2021. 22) Boker A (EUP): Marks, 1983a.
- 23) Boker BE (LMP): Marks, 1983a.
- 24) Rosh Ein Mor (LMP): Goder-Goldberger and Bar-Matthews, 2019.
- 25) N Nizzana XIII (EUP): Gilead and Bar-Yosef, 1993.
- 26) Qadesh Barnea (EUP): Goring-Morris and Davidson, 2006.
- 27) Lagama sites (EUP): Gilead, 1983. 28) Abu Noshra I & II (EUP): Phillips, 1988.
- 29) Tor Sabiha (LMP): Henry, 1995. 30) Tor Faraj (LMP): Henry, 2003.
- 31) Wadi Aghar (IUP): Kadowaki et al., 2019b. 32) Tor Fawaz (IUP): Kadowaki et al., 2022a.
- 33) Tor Hamar (EUP): Naito et al., 2022.

Chapter 2. Research materials

2.1. Lithic assemblages from the Jebel Qalkha area, Jordan

The initial investigations of the prehistoric sites in the Jebel Qalkha area were conducted by D. O. Henry as part of a long-term prehistoric project between 1976 and 1999. Systematic surveys and excavations at numerous prehistoric sites were conducted in several study areas with different elevational, topographic and biotic settings including the Mediterranean zone on the Ma'an Plateau, the Irano-Turanian steppe in the Judayid Basin, the Saharo-Arabian desert in the lowland of Wadi Hisma, and the Wadi Araba Rift Valley (Henry, 1989, 1994, 1995, 2003, 2012, 2017a, 2017b; Henry and Beaver, 2014; Henry et al., 2017). Since 2016, renewed investigation focusing on the Jebel Qalkha area has been conducted to refine the cultural-chronology and to increase human behavioral and paleoenvironmental records in the late Pleistocene (Hirose et al., 2022; Kadowaki and Henry, 2019; Kadowaki et al., 2019a, 2019b, 2021, 2022a, 2022b; Naito et al., 2022; Suga et al., 2022, 2023; Ichinose et al., 2023).

In previous studies of lithic raw materials in the Jebel Qalkha area, Henry (1995) suggested that chert raw material was transported to the MP site at Tor Faraj from sources 20 km away. Henry and Mraz (2020) also compared the use of lithic raw material between the LMP and Neolithic assemblages.

The current environment around the Jebel Qalkha area is arid with less than 50 mm annual precipitation. The vegetation is currently sparse at the transition between the Irano-Turanian and Saharo-Arabian phytogeographic zones. However, recent paleoclimatic studies suggest more humid conditions during MIS 4–3 (Torfstein et al., 2015; Miebach et al., 2019) that correspond to the Middle Paleolithic, Upper Paleolithic, and Epipaleolithic occupations in the Jebel Qalkha area. In addition, the pollen records from the dead sea indicate that a steppe corridor was widely distributed in southern Jordan (Richter et al., 2020). The analyses of pollens and phytoliths excavated from the Jebel Qalkha sites (Henry, 1995) indicate that a wider range of vegetation including arboreal and herbaceous plants was distributed during the Middle Paleolithic, Upper Paleolithic, and Epipaleolithic periods in agreement with the Dead Sea pollen records. In addition, some recent studies reported the luminescence ages in MIS 5 at wetland sediments in Wadi Gharandal (Al-Saqaret et al., 2020; Abbas et al., 2023).

This dissertation analyzes lithic assemblages collected in the renewed investigations at Tor Faraj, Wadi Aghar, Tor Fawaz, and Tor Hamar in Chapter 3 (Table 2.1.). These sites are located close to each other (less than 2 km apart) within the same geological settings characterized by extensive exposure of Umm 'Ishrin Sandstone (Figs. 2.1 and 2.2.) (Rabb'a, 1987). Chert is the dominant lithic raw material at these sites. The appearance of chert made into artifacts varies in macroscopic features, such as color and texture, and their variations are generally similar among the sites (Fig. 2.3.). Thus, I assume that prehistoric inhabitants of the sites experienced similar availability and accessibility of lithic raw materials. The following section summarizes the basic information of the four sites (Fig.2.2.).

Tor Faraj

Tor Faraj (29° 56' 19.9"N, 35° 19' 33.6" E, 1000 m a.s.l.) is a rock shelter site (Fig. 2.2e.). Several radiometric dates (TL, AAR and U-series) between 43.8 and 69.0 kya were reported (Henry, 2003). The lithic technology is characterized by the dominant employment of Levallois technology encompassing various core-flaking methods (Henry, 2003; Demidenko and Usik, 2003; Kadowaki and Henry, 2019). Especially, unidirectional-convergent reduction is dominant, but bidirectional reduction is also relatively common, and centripetal reduction is rare (Groucutt, 2014; Groucutt et al., 2015). These characteristics correspond to the lithic technology in the LMP (Hovers, 2009; Shea, 2013; Sharon and Oron, 2014; Meignen, 2019; Malinsky-Buller et al., 2021).

Wadi Aghar

Wadi Aghar (29°56'11.99" N, 35°19'53.53" E, 965 m a.s.l.) is a shallow rock shelter site (Fig. 2.2b.). The assemblage was dated to 45–40 ka by OSL and radiocarbon dating (Kadowaki et al., 2019b). The lithic assemblage includes typical UP tool types like end scrapers and burins. The core reduction technology is dominated by unidirectional flaking, and the core types are characterized by the predominance of volumetric cores with a few along-axis cores (Kadowaki et al. 2019b). In sum, the lithic assemblages from Wadi Aghar show IUP techno-typological characteristics (Kuhn and Zwyns, 2014; Nishiaki, 2018; Kadowaki et al., 2019a, 2019b, 2022a; Kuhn, 2019; Goder-Goldberger et al., 2023).

Tor Fawaz

Tor Fawaz (29°56' 49.44" N, 35°20'9.03" E, 980 m a.s.l.) is a rock shelter site (Fig. 2.2c.). The age of IUP occupations was estimated to 45–36 ka (Kadowaki et al., 2022a). The lithic assemblage includes Levallois-like points with unidirectional convergent flaking, but many prismatic blade cores and flat-faced cores are observable. The blade technology is characterized by the production of robust blades with large platforms (Kadowaki et al., 2022a). In sum, the lithic assemblages from Tor Fawaz show IUP techno-typological characteristics (Kuhn and Zwyns, 2014; Nishiaki, 2018; Kadowaki et al., 2019a, 2019b, 2022a; Kuhn, 2019; Goder-Goldberger et al., 2023).

Tor Hamar

Tor Hamar (29°56'17.34" N, 35°19'8.90" E, 985 m a.s.l.) is a rock shelter site (Fig. 2.2d.). The assemblage can be subdivided into three cultural periods: Middle Epi (Mushabian) in Layers A–E1 (15.5–15.2 ka), Early Epi (Qalkhan) in Layer E2 (24–18 ka) and Early UP (Ahmarian) in Layers F–G (38–37 ka) (Naito et al. 2022). I analyzed lithic assemblage in Layers F–G in this dissertation. The lithic assemblage from Layers F–G includes el-Wad points and is characterized by dominant production of bladelets typically from single-platform narrow-fronted cores. The techno-typological features show affiliation with the Ahmarian (Shea, 2013; Kadowaki et al., 2015; Goring-Morris and Belfer-Cohen, 2018; Richter et al., 2020; Abulafia et al., 2021; Kadowaki et al., 2021; Gennai et al., 2023; Shemer et al., 2023).

2.2. Geological settings in the Western Hisma Basin

Around the Jebel Qalkha sites, the Cambrian sandstone is dominantly exposed, and chert sources are generally limited (Rabb'a, 1987; Tarawneh, 2002, 2019). However, there are small, fragmented exposures of Cretaceous-Paleogene limestone sequences within 10 km from the Jebel Qalkha area, in which a few chert outcrops occur in small pockets (Fig. 2.1.; Kadowaki et al., 2022b; Ichinose et al., 2023). Some of the chert outcrops are distributed with numerous chert nodules and clasts suitable for the production of flaked stone tools (Fig. 2.4.). For example, the outcrop of the Miocene conglomerate at Humayma, only a few kilometers from the Jebel Qalkha area, is characterized by relatively small size of chert nodules (Fig. 2.2g). On the other hand, large chert nodules weighing over 30 kilograms are available at the Abbasiyah outcrop of Cretaceous–Paleogene limestone formations, located 6–7 km to the northeast of the Jebel Qalkha area, and at the Wadi Abu Sawwan outcrop of the Miocene conglomerate, located about 8 km to the southwest (Figs. 2.2f. and 2.2h.). In these chert sources, the survey found diagnostic Paleolithic artifacts, such as Levallois products and handaxes (Kadowaki et al., 2022b). The prehistoric exploitation of these outcrops is indicated by surface scatters of many lithic artifacts. This dissertation analyzes chert nodules collected from these outcrops in Chapter 4.



Fig. 2.1. a) Satellite image (acquired from Google Earth) of the western Hisma Basin in southern Jordan. b) Jebel Qalkha area, showing the locations of the Middle and Upper Paleolithic sites analyzed in this dissertation.



Fig. 2.2. Overviews of Paleolithic sites and chert outcrops around the Jebel Qalkha area. a) Jebel Qalkha area. b) Wadi Aghar. C) Tor Fawaz. D) Tor Hamar. E) Tor Faraj. F) Chert beds near Abbasiyah. G) A hillock strewn with chert nodules near Humayma. H) Conglomerate containing chert nodules near Wadi Abu Sawwan.



Fig. 2.3. Typical lithic artifacts analyzed in this dissertation.



Fig. 2.4. Several photographs of lithic artifacts (a–e) and chert nodules (f–k) collected in the outcrops.

Site	Tor Faraj		Wadi Aghar		Tor Fawaz		Tor Hamar		
Chrono-cultural entities	Late Mide	lle	Initial	Upper	Initial	Upper	Early	U	Jpper
	Paleolithic		Paleolithic		Paleolithic		Paleolithic		
							(Ahmarian)		
Excavation areas (Units)	A4, B2, B3, H	3 4	100, 101, C, D,		6, 7, 8, 9, 10		9, 10, 11		
			83–1, 83–2						
Layers	D2–E		B–D2		Surface, B2, C		F, G		
Date	69–44 ka		45–40 ka		45–36 ka		38–37 ka		
Core	11		13		82		63		
Fully cortical debitage	24		28		334		83		
Partially cortical debitage	76		79		673		300		
Non-cortical debitage	331		232		1670		1617		
Core trimming element	37		11		60		49		
Retouched tool	24		38		189		137		
TOTAL (without debris)	503		401		3008		2249		
References	Henry, 2003		Henry, 1995		Henry, 1	1995	Henry, 1995		
	Kadowaki a	nd	Kadowa	ıki et al.,	Kadowa	ıki et al.,	Naito	et	al.,
	Henry, 2019		2019b		2022a		2022		

Table 2.1. General inventories of lithic artifacts analyzed in this study from Tor Faraj, WadiAghar, Tor Fawaz, and Tor Hamar in the Jebel Qalkha area, southern Jordan.

Chapter 3. Diachronic changes in lithic raw material (chert) use

3.1. Methods

3.1.1. Macroscopic criteria of raw material classification

In the Levantine Paleolithic, the macroscopic attributes widely used to classify chert raw material include color, pattern, texture, cortex, luster, translucence, and inclusions (e.g, Delage, 2007; Wilson et al., 2016; Agam, 2020; Agam and Zupancich, 2020; Agam et al., 2022). In the Jebel Qalkha sites, Henry and Mraz (2020) employed a macroscopic raw material classification to compare the variability of lithic raw material between Tor Faraj (LMP) in the Jebel Qalkha area and Ayn Abū Nukhayla (Neolithic) in the Wadi Rum area. This study devised a new classification scheme suited to clarify the lithic raw material variations in the LMP, IUP and EUP assemblages of the Jebel Qalkha sites. For macroscopic examination of chert variability, several attributes were used, such as color, UV signature, texture, and translucence, which were also employed in the previous studies (Henry, 1995; Henry and Mraz, 2020). Chips and heavily weathered or burnt lithics were excluded from the analyses.

The translucency was determined as 'high' or 'low' according to the degree to which light passed through the lithic edges (> 1 mm in thickness) by placing them over a 1 cm hole backlit by the light of a 7 W LED bulb with 360 lm (NEC LED stand, HSD16002K-D12) (Fig. 3.1a.).

Colors of lithics were determined according to a Munsell Soil Color chart (e.g., Milne et al., 2009). Multiple colors were recorded for chert types with banded or gradient patterns (Fig. 3.1b.).

To obtain UV signatures of chert, I used a UV lamp (Nichia UVLED, SK375UV-002) and followed methods of Lyons et al. (2003) and Henry and Mraz (2020). The UV colors were determined according to Aubuchon's classifications (Aubuchon's, 2007), including Orange 590, Yellow 570, Violet 400, and Grey or White (Fig. 3.1c.).

Regarding the texture, this study made detailed measurements of the surface roughness as described in the next section.

3.1.2. Texture and the measurement of surface roughness (R_a)

Texture is usually recorded with ordinal scales such as fine, medium and coarse (e.g., Henry, 1995; Henry et al., 2014; Henry and Mraz, 2020). In this dissertation, a novel method was employed for the quantification of texture by measuring surface roughness (R_a values) by using Mitsutoyo SURFTEST SJ-210.

 R_a is usually used for the inspection of cut and ground surfaces of industrial products (e.g., Whitehouse, 1997; Gökkaya and Nalbant, 2007; Jeyapoovan and Murugan, 2013). Some archaeological studies also employed the measurement of surface roughness in heat treatment of lithic raw material (Schmidt et al., 2017, 2020; Schmidt and Hiscock, 2019). The measurement is taken by scratching a surface with a micro-needle (Fig. 3.1d.). The radius of the needle's tip is 5 µm (Fig. 3.1e.). To obtain the R_a value, the influence of large undulations (e.g., macroscopic curvature or ripples) is removed to focus on the roughness that corresponds to the texture of the surface (See Bhushan, 2001 for the details of the measurement).

I used Mitsutoyo SURFTEST SJ-210 and followed the measurement basis of Japanese Industrial Standard 1994. The cut-off value was λc 0.8. The measurement speed was 0.5 mm/s, and the number of intervals was x5. To determine R_a , four measurements were made for each sample by using a flat, unweathered surface, which is either ventral or dorsal in the case of artifacts. The four measurements consist of two measurements for each of two perpendicular directions. The four R_a values were then averaged to determine the R_a value of each lithic sample. The ranges of R_a values in each sample are within 0.5 µm.

 R_a values were measured for all retouched tools (n = 324) and their R_a values were compared among the four lithic assemblages to establish a quantitative criterion for assessing the surface roughness of chert artifacts. This criterion was then used to establish reference chert samples with known R_a values. The reference samples were then used to judge the texture of debitage pieces by touching them with a finger. Because the large sample size of debitage did not allow all the debitage pieces to be measured for R_a values, their texture was classified into two categories (i.e., medium and fine) by comparing their texture with the reference samples with known R_a values. The same method was applied to small lithics (e.g., retouched bladelet) that are difficult to measure with the instrument. In addition to lithic artifacts, the R_a values of chert samples were measured, which were collected from the local outcrops (n = 41) within 10 km of the sites, e.g., Humayma, Abasiyya, and Wadi Abu Sawwan.

To evaluate the validity of this new method for measuring chert texture, major chemical composition and the abundance/preservation of microfossils were also analyzed. Then, their correlations were examined with the ratio-scale data (R_a values) of surface roughness. This chapter demonstrates the validity of the new measurement of surface roughness by showing that ratio-scale data (R_a values) of the surface roughness are correlated in reasonable ways to the major chemical composition and the abundance/preservation of microfossils. Then, this chapter presents a new classification scheme designed to

characterize diachronic changes in the chert variability observed in the LMP, IUP and EUP assemblages in the Jebel Qalkha sites. The relationship is also examined between the chert variability and the techno-typological attributes of lithic artifacts.

3.1.3. Microfossil inclusion and preservation

The previous classification of chert from the Jebel Qalkha sites noted varying amounts of microfossils (Henry, 1995). The observation of microfossils can be useful for characterizing lithic raw material variations in combination with macroscopic attributes (Browne and Wilson, 2011). For the observation of microfossils, chert samples (n = 92) were selected from the lithic assemblages and outcrops around the Jebel Qalkha sites including a wide range of color, texture, and translucency. From the selected samples, thin sections were made to observe the amount and preservation of microfossils under a polarizing microscope.

Chert samples were characterized by recording "fossil scores" (0–7) that are ordinal scales based on the abundance and the preservation of microfossils in chert (see Ichinose et al., 2023 for the definition of the fossil score). The observation of microfossils in thinsections is not precise enough for the reliable identification of the species that would allow the age determination of rocks. For reliable identification, microfossils need to be isolated from chert, and this attempt is in progress as another study. Thus, this study focuses on the observations of quantity and preservation of microfossils. These records are used not for the identification of geological sources of chert but for the examination of their correlation (n = 53) with R_a values that are newly introduced as the surface roughness measurement. The analyses were conducted by Dr. Kazuhiro Tsukada (Nagoya University Museum) and Ms. Natsuki Ichinose (Graduate School of Environmental Studies, Nagoya University).

3.1.4. Measurement of major chemical composition

The chemical composition of chert was measured using an Energy Dispersive X-ray spectrometer linked with the Scanning Electron Microscope (SEM-EDX) to examine its correlations with R_a values, a new measurement for chert texture. For this purpose, chert samples (n = 33) were selected from the Tor Fawaz assemblage, including a wide range of color, texture, and translucency. Multiple measurements were made on different spots for each sample and used the mean value to average the internal variation of chemical composition (SEM, Hitachi S-3400 N). The oxide concentration was used for comparison, and various components were detected, such as SiO₂, CaO, P₂O₅, Al₂O₃, K₂O, Na₂O, and MgO. Their total amount was normalized as 100% following Tsukada (2018). The ranges

of major chemical components in each sample are within 5%. The detected components were dominated by SiO_2 and CaO while the other components accounted for less than 1.5%. I focused on the major components, i.e., SiO_2 and CaO, to examine their correlations with the surface roughness (R_a values). The Mann–Whitney U test was conducted to evaluate whether the chemical composition differs significantly between the visual chert types.

3.1.5. Techno-typological comparison among the chert types

The chert classifications, as established by the methods above, were used to examine diachronic changes in the use of lithic raw material in the LMP, IUP, and EUP assemblages of the Jebel Qalkha sties. The relative amounts of the chert types were calculated by the number and the weight of lithics. In addition, I examined whether the chert types established in this study show any correlation with flaking methods or lithic morphologies (e.g., Wilson et al., 2016; Agam, 2020; Sánchez de la Torre et al., 2020). For this purpose, I compared techno-typological attributes, particularly several techno-typological indices used in some previous studies (Inizan et al., 1999; Tostevin, 2012; Kot et al., 2020; Abulafia et al., 2021), among the chert types.

3.2. Results

3.2.1. Classification of chert types based on macroscopic attributes

Based on the macroscopic attributes, the chert artifacts were classified into ten raw material types (Fig. 3.2.; Table S1 in Appendix). I employed a method of taxonomic classification (Banning, 2020) and used texture as a top criterion for classification because it is the most quantitative and objective attribute among the macroscopic attributes used in this study. I identified two textural categories: medium and fine. A criterion for this division was established from the distribution of the surface roughness values (R_a) measured on retouched tools (Fig. 3.3.). The distributions of R_a values are similar among Tor Faraj, Wadi Aghar and Tor Fawaz while that of Tor Hamar is distinct with a mode at 1.5–2 in R_a values. Based on this observation, R_a value 2 was set as a boundary between medium and fine textures. The geological chert samples from the local sources showed a similar range of R_a values to those of the lithic assemblages in the Jebel Qalkha sites. Moreover, the distribution is similar to those of the Tor Faraj, Wadi Aghar and Tor Fawaz assemblages.

Fine texture lithics were subdivided by the variation of translucency, for which were identified into two sub-categories, i.e., low and high translucency. Each of the sub-categories was further divided by color variations. All medium texture lithics showed low translucency. Thus, they were subdivided by color variations and then UV signatures.

While ten chert types were identified, as described above, their correlations with petrographic and geochemical attributes suggested three major type-groups (Types M, FL and FH). The following describes how the three type-groups are related to petrographic and geochemical characteristics.

- Type M is the first group consists of all the chert artifacts with medium texture. Its color varies from brown to orange and shows various patterns of color bands and gradients. Because both color and its pattern vary gradually within individual samples, they cannot be effective criteria for the subdivision of Type M chert. Cortex is thin (< 1 mm) and worn, and its color is white to brown.
- Type FL is the second group includes the chert artifacts with fine texture and low translucency. The color variations do not differ from those of Type M, but Type FL characteristically includes white and black colors. The variety of cortex is also similar to Type M chert.

 Type FH is the chert in the third group also has fine texture but has high translucency. Two major color groups in Type FH are dark brown (Dk. Dusky, Red) and light brown (Red Grey, Weak Red). The cortex is thinner than that of Type M. Some samples show white surface layers of 1–2-mm thickness, which likely resulted from weathering called "white alteration" (Thiry et al., 2014; Caux et al., 2018).

3.2.2. Petrographic and geochemical analysis

According to the microscopic observations of thin-sections of the lithics, Type FH generally includes only a few microfossils, while Type M includes numerous microfossils. Type FL is intermediate in microfossil abundance between Type FH and Type M (Fig. 3.4.). In addition, microfossils in Type FH and Type FL are poorly preserved or merged with the matrix, those in Type M are, on the contrary, preserved well enough to show clear skeletal structures.

More quantitatively, Fig. 3.5. shows the relationship between microfossil abundance/preservation and the texture of chert. The surface roughness (R_a) and fossil scores generally show a positive correlation. This suggests that chert with greater surface roughness, i.e., higher R_a values, tends to include a greater amount and preservation of microfossils. The Spearman's correlation coefficient is 0.518 (p-value <0.001).

Fig. 3.6. shows the relationship between the texture of the lithics (R_a values) and SiO₂ concentration for the samples. The two attributes show a negative correlation. This means that chert artifacts with a medium/coarse texture (higher R_a values), i.e., Type M, tend to be lower in SiO₂ concentration than those of finer texture, i.e., Types FL and FH. The Spearman's correlation coefficient between them is 0.523 (p-value = 0.002).

3.2.3. Diachronic trends and techno-typological characteristics by the chert types

Fig. 3.7. shows the relative frequencies of the three type-groups of chert (Types M, FL and FH) in the four lithic assemblages, representing the LMP, IUP and EUP periods. The LMP (Tor Faraj) and IUP (Wadi Aghar and Tor Fawaz) assemblages are characterized by a greater proportion of Type M than the EUP (Tor Hamar) that instead shows a higher proportion of Type FH. Type FH accounts for nearly 50% of the Tor Hamar assemblage. The increase of Type FH from the IUP to the EUP is statistically significant (chi-square value = 21.883, p-value <0.001 between Tor Fawaz and Tor Hamar; chi-square value = 11.554, p-value = 0.009 between Wadi Aghar and Tor Hamar). Thus, the EUP can be differentiated from the IUP/LMP in the decrease of Type M and the concomitant increase of Type FH. Based on this observation, I focus on the comparison between Type M and Type FH in the following results.

To examine whether Type FH chert was associated with any specific lithic technomorphological categories, the lithic assemblages from the four sites were divided into six general categories, including cortical debitage, partially cortical debitage, non-cortical debitage, core trimming elements, cores, and retouched tools. Then, relative frequencies of the six lithic classes were compared between Type M and Type FH for each of the sites (Fig. 3.8.). The results show that the EUP assemblage (Tor Hamar) is characterized by higher proportions of non-cortical debitage than the LMP/IUP assemblages. Although this observation applies to both Types M and FH, Type FH shows a slightly greater proportion of non-cortical debitage than Type M.

Fig. 3.9. shows the relative frequencies of five blank morphologies, including blade, flake, point, spall and bladelet, by Type M and Type FH in the four sites. The EUP site is characterized by greater ratios of bladelets than the LMP/IUP in both of Type M and Type FH, reflecting the development of bladelet technology (Kadowaki et al., 2021). On the other hand, the four sites commonly show greater proportions of bladelets in Type FH than those in Type M. The proportion of bladelets in Type FH is particularly high at Tor Hamar, accounting for 43.44%.

Table 3.1. compares Type M and Type FH in several measurements of complete artifacts from the four sites. The measurements include length, width, thickness, weight, elongation, and flattening. Because all the measurement data are not distributed normally, the Mann–Whitney U test was conducted to assess the significance of difference between Type M and Type FH. Length, width, thickness and weight of Type M are generally greater than those in Type FH in each of the lithic assemblages. In addition, lithics made on Type FH are generally more elongated than those made on Type M although this difference is statistically significant only in the Tor Hamar assemblage. As for flattening, no significant difference was observed between Type M and Type FH. These patterns are the most prominent in the EUP assemblage at Tor Hamar. Collectively, lithics in Type M chert tend to be larger than those in Type FH in each of the LMP, IUP, and EUP assemblages. Additionally, some other techno-typological attributes were also compared, such as retouched tool types, core morphology, dorsal scar patterns, dorsal-distal shapes, platform types, and overhang removal, among Types M, FL, and FH. However, no clear difference was observed among the three chert type-groups.

3.3. Discussion

3.3.1. Three chert type-groups, M, FL, and FH

As shown in the results, this study identified ten raw material types using macroscopic criteria of texture, translucency, color and UV signature (Fig. 3.2.). The texture was quantified through the measurement of R_a values, which showed reasonable and expectable correlations with major chemical components (SiO₂ and CaO) and the abundance/preservation of microfossils in chert (Figs. 3.5 and 3.6.). Thus, I suggest that R_a values can be reliable ratio-scale data for the surface roughness of chert.

Translucency was evaluated as the degree of fixed light passing through the edges of lithics. In contrast, I found color variations more difficult to define consistently as they gradually change and vary even in the same sample. Additionally, UV signature may be influenced by the degree of weathering although heavily weathered surfaces were avoided. As a result, clear diachronic trends could not be observed in the relative frequencies of the ten chert types. Thus, I focused on texture and translucency as principal criteria in the taxonomic classification of chert types because they were based on more quantitative and objective measurements. These attributes were likely recognizable to the prehistoric inhabitants in the Jebel Qalkha area. In fact, a clear diachronic trend was observed in the three type-groups (Types M, FL, and FH) as described in the Result section (Chapter 3.2.3.). This shows that the three chert type-groups are sufficient to highlight diachronic changes in the use of lithic raw materials over the LMP, IUP, and EUP.

In this study, I did not examine whether the three type-groups of chert represent specifically different geological sources. However, it is at least clear that all three chert type-groups are locally available within the 10 km radius of the Jebel Qalkha area, as shown by the distribution of R_a values of the chert samples (Fig. 3.3.). Thus, I suggest that there was no significant difference in availability among the three chert type-groups for the prehistoric inhabitants in the Jebel Qalkha area. This indicates that the diachronic changes in the proportions of the three chert type-groups were related to changes in the selection of lithic raw materials according to lithic technology that changed over the LMP, IUP, and EUP, as discussed in the following sections.

Further research needs more geologic chert samples from potential sources around the sites and needs to examine their optical, surface profile, petrographic, and geochemical information. Comparison with these geological data can help us establish the chert classification related to geological sources and will allow us to discuss raw material provisioning behaviors (e.g., Browne and Wilson, 2011; Agam, 2020).

3.3.2. Increase of type FH chert in the EUP

A clear change in the relative frequencies of the three chert type-groups (i.e., Types M, FL, and FH) was detected in the EUP, which showed a higher frequency of Type FH than the IUP and LMP. The increase of Type FH in the EUP could possibly reflect a change in chert sources. However, according to the geological surveys around the Jebel Qalkha area, the variability of cherts at several local outcrops includes Types M, FL, and FH. This is illustrated by a wide range of the surface roughness (R_a values) of chert samples from outcrops in the Jebel Qalkha area (Fig. 3.3.). Thus, Type FH does not necessarily correspond to a specific chert source. On the other hand, this is indicated by differences in chemical compositions of some EUP chert artifacts from those of the LMP and IUP in the Jebel Qalkha area (Ichinose et al., 2023). However, the possibility of changes in raw material sources is not exclusive from the intentional selection of Type FH chert that was suitable for the production of bladelets.

On the other hand, the increase of Type FH chert in the EUP assemblage from Tor Hamar could have been caused by an increased selection for Type FH chert at outcrops. One of the reasons for favoring Type FH may be related to the production of bladelets that increased in the EUP (Kadowaki et al., 2021). As shown in Fig. 3.9., more bladelets are made on Type FH than Type M not only in the EUP but also in the IUP and LMP. In addition, lithic artefacts made on Type FH chert material are generally smaller than those on Type M in all the periods (Table 3.1.). Thus, I suggest that small lithic artifacts, particularly bladelets, tended to be produced from Type FH chert. Importantly, the association between Type FH chert and small lithics applies to all three periods despite diachronic changes in lithic technology, and the increase of Type FH in the EUP was clearly associated with the development of bladelet technology and concomitant decrease in the size of lithics in the EUP. In Chapter 4, the more detailed discussion based on the mechanical properties is described.

3.3.3. Reasons for the association between type FH chert and bladelets

The development of bladelet technology in the EUP can be evaluated as a phenomena of "lithic miniaturization" in which the production of small stone tools is a dominant knapping mode in lithic assemblages (Shea and Pargeter, 2019; Shipton, 2023). In addition, these studies suggested that miniaturized stone tools include several functional advantages like compound tools, arrow tips, scarification, and shaving. The increase of Type FH chert in the EUP is likely to have been related to the concomitant lithic miniaturization in general. One of the possible reasons for this association is the size of

chert raw materials. Fig. 3.10. shows the scatterplot of length and width of cobbles/nodules (n = 38) that were sampled at chert outcrops around the Jebel Qalkha area. The size of cobbles/nodules, represented by length multiplied by width, is significantly greater in Type M than Type FH (p-value of the Mann–Whitney U = 0.006).

Given this size difference, the larger volume of Type M chert is likely to have been more suited than Type FH for Levallois core reduction in the LMP and the production of robust blades in the IUP. On the other hand, the production of bladelets in the EUP may not have required as large cobbles or nodules but could use a wider range of size, including smaller cobbles of Type FH. In this way, the increase of Type FH chert in the EUP is likely to reflect a wider range of chert selection, in terms of size, for the bladelet production. However, the dimensional data of chert cobbles/nodules at the outcrops are based on a small samples size that needs to be verified with more systematic collections in future.

On the other hand, it is also possible that Type FH was favored in the EUP because of its fine texture in relation to bladelet production. This idea is consistent with some patterns in the Epipaleolithic raw material use in the Levant, where fine-grained flint or chalcedony was preferred for the production of microliths whereas larger tools, such as scrapers, were made on coarse-grained cherts (Henry, 1989; Delage, 2007). Marder and Goring-Morris (2020) have also reported preferential use of translucent chalcedony in the Epipaleolithic assemblages in the Negev and suggested less brittleness of chalcedony as one of possible reasons for the preference. Assuming that 'fine-grained chert' and 'chalcedony' are similar to chert with fine texture in this study, these observations indicate that the association between cherts with fine texture and small lithics, such as bladelets and microliths, is a broader phenomena in the Upper and Epipaleolithic periods in the Levant. Chapter 4 will analyze of physical characteristics between fine and medium-coarse texture chert and delve into the reasons for this association.

3.4. Summary

This chapter presented a rare study of changes in lithic raw material use at the MP-UP transition in the Levant. This study took advantages of a unique condition in the Jebel Qalkha area, southern Jordan, where the LMP, IUP, and EUP sites are located close to each other, sharing essentially the same availability and accessibility of lithic raw materials.

The raw material (chert) variations were analyzed in four lithic assemblages from the Jebel Qalkha sites. While the chert variations were classified into ten types on the basis of several macroscopic attributes, more general type-groups (Types M, FL and FH) were established by focusing on two criteria (i.e., texture and translucency) that were established objectively through quantification. Differences in texture and translucency of chert were likely recognizable to prehistoric inhabitants in the Jebel Qalkha area. The relative frequencies of the three type-groups in the four assemblages in the Jebel Qalkha sites showed an increase of Type FH in the EUP assemblage (Tor Hamar).

This result means that a clear change in lithic raw material was observed not at the classical boundary between the MP and UP, but between the IUP and EUP. The increase of Type FH in the EUP was clearly associated with the development of bladelet technology and concomitant decrease in the size of lithics in the EUP. Because the varieties of locally available chert near the Jebel Qalkha area include all of Types M, FL, and FH, the increase of Type FH in the EUP may not reflect a fundamental change in chert sources. Instead, it is more likely to reflect a change in the selection of chert, possibly related to the size of nodules/ cobbles or the fine texture. This interpretation needs to be verified with other EUP lithic assemblages like Tor Aeid and Jebel Humeima in the Jebel Qalkha area (Henry, 1995).

Although this study did not detect a clear change in lithic raw material between the LMP and IUP, it does not necessarily mean that no change occurred in lithic raw material procurement or use over these periods. Any change in chert variations could be detectable by using other classification methods, such as employing an RGB color scale or identifying the chert sources more specifically. Thus, further studies are necessary to clarify the changes in lithic raw material and their behavioral implications at the MP-UP transition in the Levant.



Fig. 3.1. Images showing the methods and instruments used for the measurements of macroscopic attributes of chert artifacts. (a) Translucency. (b) Munsell Soil Color Chart. (c) UV color signatures. (d) Device for measuring average surface roughness (Mitsutoyo SURFTEST SJ-210). (e) Microneedle for measuring average surface roughness. The picture shows the R_a values of a pencil as a reference.



Fig. 3.2. A taxonomic classification of chert artifacts from Tor Faraj, Wadi Aghar, Tor Fawaz, and Tor Hamar assemblages on the basis of several macroscopic attributes.



Fig. 3.3. Distribution of R_a values of chert artifacts from the four sites in the Jebel Qalkha area and geological local chert samples from nearby chert outcrops (within 10 km of the sites). Relative frequencies of retouched tools and chert samples are shown by intervals of 0.5 in the R_a value.



Fig. 3.4. Photomicrographs of several chert artifacts from the Jebel Qalkha sites under a polarizing microscope (all photos are under crossed polars). The selected artifacts represent the three chert type-groups (Types M, FL, and FH) identified in this study. Note differences in the abundance of microfossils among the three chert type-groups.



Fig. 3.5. R_a values (µm) vs. fossil scores diagram for the lithic samples (n = 53) from the Jebel Qalkha sites by the three chert type-groups (Types M, FL, and FH).



Fig. 3.6. SiO₂ vs. R_a values (µm) diagram for the chert samples (n = 33) from the Tor Fawaz lithic assemblage. The plots are marked differently according to the three chert type-groups (Types M, FL, and FH).



Fig. 3.7. Relative frequencies of the three chert type-groups (Types M, FL, and FH) in the five lithic assemblages in the Jebel Qalkha area.



Fig. 3.8. Relative frequencies of six general categories of lithic artifacts by the two chert type-groups (Type M and Type FH) in the four lithic assemblages in the Jebel Qalkha area.


Fig. 3.9. Relative frequencies of five blank morphologies, including blade, flake, point, spall and bladelet, by the two chert type-groups (Type M and Type FH) in the four assemblages in the Jebel Qalkha area.



Fig. 3.10. Scatterplots of length and width of cobbles/nodules (n = 38) sampled at chert outcrops located around the Jebel Qalkha area. The plots are marked differently according to the three chert type-groups (Types M, FL, and FH).

	Tor Faraj (LMP)		Wadi Aghar (IUP)		Tor Fawaz (IUP)		Tor Hamar (EUP)					
	Type M		Type FH	Type M		Type FH	Type M		Type FH	Type M		Type FH
	n = 146		n = 47	n = 69		n = 37	n = 489		n = 146	n = 263		n = 367
Length (mm)	34.01		33.41	49.89		43.00	42.83	>	35.50	31.05	>	29.45
Width (mm)	26.58	>	24.15	27.80	>	21.02	25.19	>	22.48	18.05	>	14.42
Thickness (mm)	4.79	>	3.79	7.26		5.97	8.19	>	6.86	4.83	>	3.72
Weight (g)	4.27		3.07	10.25	>	5.19	8.47	>	5.49	2.81	>	1.62
Elongation	1.34		1.52	1.84		2.12	1.68		1.64	1.92	<	2.29
(Length/Width)												
Flattening	5.34		5.10	3.62		3.38	3.11		3.08	3.50		3.79
(Width/Thickness)												

Table 3.1. Median values of dimensional and morphological measurements of complete lithics from the four sites by the two chert typegroups (Type M and Type FH). '>' means statistically significant difference indicated by the Mann–Whitney U test.

Chapter 4. Quantitative examination of chert mechanical properties and fracture predictability

4.1. Background of fracture predictability

Given the fracture mechanics of flaked stone tools, percussion of vitreous rocks is understood as cracks similar to glass. Some glass engineering experiments suggest that some kinds of cracks could have been observed when fracture stress is applied to glass (Cook and Pharr, 1990). Among these cracks, the intentional production of conical (conchoidal) crack, called Hertz cone, detached flaked stone tools (Speth, 1972; Cotterell and Kamminga, 1987; Odell, 2004; McPherron et al., 2020; Lawn, 2021). There are two different cone cracks: an inner one, occurring simultaneously with the impact loading, and an outer one, produced shortly after release of impact loading (Ligkovanlis, 2022). In fact, the two cone cracks can be observed on the ventral surface of stone tools. Although the detailed mechanism remains unclear, Hertz cone is widely recognized as explaining the function of pressure and striking in flaking stone tools by archaeologists. This chapter will discuss the difference in fracture force between lithic raw material types.

In archaeology, the "quality" of lithic raw material is often evaluated on the degree of its easiness to be flaked. In numerous studies of lithic raw material, an issue of raw material "quality" has often been discussed. The "quality" of raw material can be an important basis in the selection of raw material. Besides, the raw material "quality" is used for establishing the models that quantify attractiveness of raw material sources and estimate kinds of lithic tools produced (e.g., Andrefsky, 1994, 2009; Wilson, 2007; Browne and Wilson, 2011). On the other hand, most of the discussions on raw material "quality" were qualitative. Flaked stone tools are often made of sedimentary and volcanic rocks, such as obsidian and chert, which are rich in silica content and are microcrystalline and glassy. A common recognition among archaeologists is that these properties increase the "quality" of lithic raw material (e.g., Bradbury et al., 2008; Sharon, 2008). The "quality" of chert raw material is usually evaluated on the basis of macroscopic characteristics. More specifically, archaeologists often evaluated that fine-grained (textured) chert is high quality, and medium/coarse-grained (textured) is low quality (Goring-Morris and Davidson, 2006; Delage, 2007; Neeley, 2007; Bustillo et al., 2009; Gómez de Soler et al., 2020b; Marder and Goring-Morris, 2020; Parow-Souchon and Purschwitz, 2020; Tomasso and Rots, 2021).

Several studies use the term "fracture predictability" based on the degree to which the flaking outcome can be predicted for a specific type of lithic raw material (Braun et al., 2009; Eren et al., 2014; Moník and Hadraba, 2016; Caruana and Mtshali, 2018; Egeland

et al., 2019; Williams et al., 2019; Lewis et al., 2022). The flakes knapped from lithic raw materials with the high fracture predictability (high quality) tend to have sharp edges suitable for knives or scrapers (Harmand, 2009; Terradillos-Bernal and Rodríguez-Álvarez, 2017). Some previous studies from the viewpoint of fracture predictability examined several physical attributes including brittleness, elasticity, isotropy, homogeneity, continuity, and granularity (Cotterell and Kamminga, 1987; Luedtke, 1992; Whittaker, 1994; Inizan et al., 1999; Sharon, 2008; Bamforth, 2009; Eren et al. 2011; Goldman-Neuman and Hovers, 2012; Garvey, 2015; Moník and Hadraba, 2016; Rodríguez-Rellán, 2016; Herrero-Alonso et al., 2021; Namen et al., 2022). These attributes can be classified into two group: a group of structural properties (e.g., isotropy, homogeneity, continuity and granularity) and a group of mechanical properties (e.g., brittleness and elasticity).

As for the structural properties, the high fracture predictability was linked to some petrographic characteristics, such as little or no crystalline macrostructure, few impurities (e.g., organic inclusions, void space, secondary quartz, and phenocrysts), and an overall small size of crystals (Luedtke, 1992; Domanski et al., 1994; Braun et al., 2009; Eren et al., 2014; Egeland et al., 2019). These characteristics were considered to facilitate fracture propagation inside the rock with little interference and thus increase the predictability of flaking, resulting in the feeling of "easiness" in stone knapping (Cotterell et al., 1985; Cotterell and Kamminga, 1987; Brantingham et al., 2000; Bamforth, 2009; Braun et al., 2009). Moreover, some studies approached the fracture predictability through petrographic analyses by observing macroscopic attributes and microscopic features in thin section (Brantingham et al., 2000; Stout et al., 2005; Sherwood et al., 2018). These studies aimed to quantify the fracture predictability by calculating the average size of phenocrysts and the ratio of impurities in flaking removals. Recently, some studies quantified the amount of impurities within chert specimens through loss on ignition me (LOI) method (William et al., 2019; Lewis et al., 2022).

As for the mechanical properties, several studies aimed to quantify the fracture predictability of heat-treated raw material by measuring their mechanical properties (Domanski and Webb, 1992, 2007; Domanski et al., 1994; Domanski et al., 2009; Yonekura, 2010; Schmidt et al., 2012, 2019; Zhou et al., 2014; Mraz et al., 2019; Moník et al., 2021; Nickel and Schmidt, 2022). Some studies explained that the heat-treatment of silica rocks increase their quality due to the formation of new Si–O–Si bonds that make the fracture path less meandering (Schmidt et al., 2012, 2019). In addition to heat treatment, some studies examined the fracture predictability of different kinds of lithic raw material by comparing mechanical properties (Goodman, 1944; Webb and Domanski,

2008; Tsobgou and Dabard, 2010; McPherron et al., 2014; Moník and Hadraba, 2016; Rodríguez-Rellán, 2016; Namen et al., 2022). These studies mainly measured strength, elasticity and toughness as key indicators. These mechanical properties are used not only in petrology but also in materials science. However, in their application to lithic studies in archaeology, it is important to be aware that most methods for measuring mechanical properties of tool stones do not exactly replicate actual stone knapping (e.g., angles of force). In addition, only a few studies present explicit models that explain how the mechanical properties influence stone knapping and lithic production (Nickel and Schmidt, 2022). The following section summarizes current archaeological views on how strength, elasticity, and toughness are related to stone knapping and flaking phenomena.

Strength Strength is the absolute value of fracture stress when objects break down. Lithic raw materials with high strength can be useful in its use for producing flakes as it prevents shattering of flakes' striking platforms (Doelman et al., 2001). On the other hand, a previous study pointed out potential fluctuations in the strength value of brittle materials with internal cracks and indicated an importance of using appropriate kinds of tests and sufficient numbers of rock samples (Tsirk, 2014). More specifically, many studies employed compressive or tensile strength as parameters of lithic raw material strength (Domanski et al., 1994; Webb and Domanski, 2008; Yonekura and Suzuki, 2009; Zhou et al., 2014; Nickel and Schmidt, 2022).

Elasticity Elasticity is the characteristic of objects returning to the original form after their plastic deformation. Lithic raw materials behave elastically at a macroscopic scale, and elasticity is a useful index that represents flexibility. Previous studies quantified elasticity by measuring Young's modulus (Domanski et al., 1994; Schmidt et al., 2019; Moník et al., 2021; Namen et al., 2022). The higher value of Young's modulus means greater resistance to deformation (Luedtke, 1992; Braun et al., 2009).

Toughness Toughness is the degree of resistance of a material to crack opening, and this characteristic is used for brittle materials with internal cracks. Previous studies quantified toughness by measuring fracture toughness (K_{IC}) (Domanski and Webb, 1992; Domanski et al., 1994; Webb and Domanski, 2008; Domanski et al., 2009; Schmidt et al., 2019; Moník et al., 2021; Namen et al., 2022; Nickel and Schmidt, 2022). Raw materials of flaked stone tools, such as obsidian and chert, are similar to glass, and they often have microscopic cracks inside the rocks. When these internal cracks receive certain load (fracture stress), they induce a sudden breakage of the material. Fracture toughness is considered a useful parameter to evaluate raw materials of flaked stone tools (Tsirk, 2014).

Geological studies suggest that the above three mechanical properties are correlated mutually in general (Sachpazis, 1990; Zhang, 2002; Yasar and Erdogan, 2004; Aydin and Basu, 2005). In addition, the mechanical properties can be discussed in relation to the petrographic characteristics (Namen et al., 2022). Here I summarize some previous studies' predictions about how the mechanical properties of rocks influence stone flaking. Firstly, high strength and elasticity of tool stones are considered to facilitate the propagation of fracture forces and thus allows for predictable flaking (Braun et al., 2009; Egeland et al., 2019). Secondly, low fracture toughness makes rocks to be flaked with less forces, facilitating the preparations of core ridges and platforms (Doelman et al., 2001). On the other hand, when fracture toughness (K_{IC}) is too large, the raw material is too tough to work (Tsirk, 2014). The lithic grade scale proposed by Callahan (1979) suggests that lithic raw materials of better workability (high fracture predictability) have large values of elasticity and modest values of toughness (KIC). Recently, Nickel and Schmidt (2022) proposed a mathematical model in which a combination of low fracture resistance (K_{IC}) and high tension stress reduces the size of a critical crack required to induce stone flaking, i.e., "Griffith" crack length, leading to high knapping quality of rocks.

Based on the previous studies, I estimate that lithic raw materials with high fracture predictability are rocks with high strength and elasticity but low toughness. To evaluate some of these mechanical properties, this study focuses on two types of properties that, are explained below.

4.2. Methods

4.2.1. Schmidt Hammer

Schmidt Hammer (N-type) is the instrument for measuring rebound hardness (Q-value). Despite the term of "hardness", the measurement by Schmidt Hammer is also related to elastic behavior of material because it measures the distance of elastic rebound applied under a controlled impact (2.207 Nm) on a rock surface. A spring-loaded piston makes an impact onto a plunger that transfers a force to a rock surface (Fig. S1 in Appendix). This measurement mechanism is somewhat similar to the indirect percussion in stone knapping. Schmidt Hammer has been used in some archaeological studies to compare the rebound hardness among several lithic raw materials (Braun et al., 2009; Eren et al., 2014; Egeland et al., 2019).

Schmidt Hammer is used in some engineering and geological studies to measure strengths of buildings and roads, and degrees of weathering of rock surfaces. The geological studies explained the measurement mechanism of Schmidt Hammer in detail (Betts and Latta, 2000; Goodie, 2006; Aydin, 2009; Winkler and Matthews, 2014; Ghorbani et al., 2022; Matthews and Winkler, 2022). As for the mechanical properties above-mentioned, many geological studies suggested that rebound hardness is positively correlated with strength (compressive strength) and elasticity (Young's modulus) (Winkler, 1975; Katz et al., 2000; Yasar and Erdogan, 2004; Aydin and Basu, 2005; Goudie, 2006; Yagiz, 2009; Bilen, 2021; Teymen, 2021).

In particular, the relationship with elasticity can be understood as follows. A rock with high elastic modulus is less deformable, and it rebounds with less absorption of the impact, resulting in higher rebound hardness. In fact, Callahan (1979) suggested that the high elastic modulus of lithic raw materials facilitates the control of flaking. Some archaeological studies used Schmidt Hammer to measure rebound hardness measured as a proxy for the fracture predictability because lithic raw materials that are often recognized easy to knap, like obsidian and chert, tend to show high rebound hardness (Braun et al., 2009; Brown et al., 2009; Eren et al., 2014; Egeland et al., 2019).

The measurement (Q-value) of rebound hardness with Schmidt Hammer can be performed quickly. All procedure can be carried out in the field. Moreover, the preparation of raw material samples is not required. The Schmidt Hammer measurement (Silver Schmidt OS8200 N) was performed under the following conditions and procedures (Table 4.1. and Fig. 4.1.) by referring to methods in previous studies in material sciences.

- 1. Chert nodules of a certain size (> 4 kg) were collected from the outcrops in the study area. The smallest dimensional limit was set in order to reduce fluctuations, errors and outliers of measurement values (c.f., Demirdag et al., 2009).
- 2. Size and mass of chert samples were recorded prior to the measurement of rebound hardness. Chert samples were classified into the chert types (Types M, FL and FH) based on the appearance and the texture of fractured surface.
- 3. Flakes removed from the nodules were collected as samples to be studied in the laboratory. R_a values and translucency were measured at the laboratory, and the chert samples were classified into the chert types by the methods of Chapter 3.
- 4. Ten measurements (Q-values) were obtained for each chert sample, and the average of the largest five Q-values was calculated and used for the comparisons (c.f., Anikoh et al., 2015; Brown, 1981).
- 5. As comparative samples, the rebound hardness of obsidian was measured, which is a typical raw material suited for the production of flaked stone tools. Obsidian nodules were obtained from Shirataki, Japan. As another kind of comparative samples, the rebound hardness of andesite was also measured, which is not suited for flaking but often used for ground stone tools. Archaeological materials of grinding querns made of andesite, were obtained from Hokkaido, Japan.

4.2.2. Rockwell Hardness (HRC)

Rockwell Hardness (HRC) is another kind of parameters for hardness. It is a static method and an indentation test in contrast to Schmidt Hammer that is a dynamic method and measurement of elastic rebound (Ghorbani et al., 2022). In the HRC measurement, a conical diamond indenter is pressed against an object twice: the first time with a smaller load (10 kgf, 98.07 N) and the second time with a large load (150 kgf, 1471 N), and a difference between the two depths of penetrations is used to determine the hardness (Fig. S2 in Appendix). Higher HRC values represent greater hardness (i.e., resistance against deformation). This measurement is mainly used for metal materials. The detailed mechanism is explained in geological and materials science studies (Broitman, 2017; Ghorbani et al., 2022).

In archaeological studies, the measurement of Rockwell Hardness was applied to pottery (Simon and Coghlan, 1989) and hard hammers used for lithic production (Magnani et al., 2014). Besides, some archaeological studies employed other indentation hardness tests for measuring lithic raw materials (Yonekura and Suzuki, 2009; Tsobgou and Dabard, 2010; Schmidt et al., 2019; Moník et al., 2021; Namen et al., 2022). The procedure of Rockwell hardness test is less complicated than those of other indentation hardness tests.

Rockwell Hardness is measured by deforming samples in a small scale. When it is applied to metals, it causes their plastic deformation. However, I expected that brittle materials, like obsidian and chert, would behave differently from metals. To confirm this idea, microscopic observations of indented areas were made on the samples after Rockwell Hardness tests.

In the measurement, Mitsutoyo ATK-600 (installed at the Graduate School of Science/School of Science Equipment Development Support Section, Nagoya University) was used under the following conditions and procedures (Table 4.1. and Fig. 4.2.).

- Chert nodules from the outcrops were cut and polished into slabs (20 mm in thickness). Two to three slab samples were made from one nodule. 16 slab samples were measured in total.
- 2. In the measurement, a conical diamond indenter was pressed perpendicularly to a flat surface of the slab sample.
- 3. Ten measurements were made for each of the slab sample to deal with the variations of measurement values.
- 4. In addition to two chert types (Type M and Type FH), obsidian (from Shirataki, Japan) and a glass block (length: 70 mm, width: 40 mm, thickness: 35 mm) were also measured as comparative samples.

4.3. Results

4.3.1. Schmidt rebound hardness

Fig. 4.3. shows the distributions of the average Q-values, each of which was obtained from each rock sample. Shapiro-Wilk tests show that these values are normally distributed in the four raw material types. The Q-values of the andesite samples are distinctively lower than the other raw material types. In contrast, the values of the other types (Type FH chert, Type M chert, and obsidian) overlap with each other within a narrow range of 73–85, which are close to the values reported for flint and obsidian in previous studies (Eren et al., 2014; Egeland et al., 2019).

The t-test does not indicate significant difference in Q-values among Type M chert, Type FH chert and obsidian (Table 4.2.). However, it is notable that obsidian shows the highest mean value, which is followed by Type FH and then Type M.

As for the distributions of Q-value (Fig. 4.4.), those in Type FH chert are narrower than Type M chert. The Kurtosis of Type FH chert (-0.424) is larger than Type M chert (-0.873) meaning that the values of Type FH chert tend to be concentrated in a narrower range of than those of Type M chert.

The relationship was also examined between the Q-value and the mass of raw materials (Fig. 4.5.) and conducted regression analyses for Type M chert, Type FH chert and obsidian respectively. Table 4.3. summarizes linear regression equations, R² values, and p-values for each lithic raw material. The R² values of Type M chert and Type FH chert are high, and the p-values are below 0.05. Thus, the linear regressions of the two chert types are reasonable. The y-intercepts of the regression lines for Type M chert and Type FH chert differ from each other by more than 3 in Q-values. On the other hand, the R² value of obsidian is low, and the p-value is higher than 0.05, probably due to the small sample size. The y-intercept of the regression line for obsidian is close to Type FH chert, and the slope of the regression line is close to Type M chert.

4.3.2. Rockwell Hardness

After the Rockwell Hardness test, indented spots on the glass block showed conchoidal fractures that are apparently similar to flaking scars created by stone knapping (Fig. 4.6.). Below the indent, two vertical semicircular fractures were visible inside the glass. These fractures clearly differ from plastic deformation of metals. The indented areas of chert and obsidian also showed conchoidal fractures similar to those of the glass block, indicating that these materials experienced similar cracking processes.

Fig. 4.7. shows the distributions of Rockwell Hardness values. Shapiro-Wilk tests show that a normal distribution applies to the measurement data of only Type M chert. The values of Type FH chert are distributed more widely than those of Type M. Besides, the values of obsidian are distributed more widely than Type FH chert. As for the medians of Rockwell Hardness values, Type M chert has the highest value (40.7), followed by Type FH chert (27.2), and the obsidian value is noticeably lower (7.9). As a result of Mann-Whitney's U test, the three lithic raw materials differ from each other significantly (Table 4.4.).

4.4. Discussion

4.4.1. Schmidt rebound hardness

As a result of Schmidt Hammer measurement, andesite of grinding querns showed low rebound hardness values. Grinding tools are often made from porous or inhomogeneous raw materials that expose rough surfaces suitable for pulverizing materials. The andesite specimens used in this study typically include crystallites of various sizes that make the rock less stable in response to the percussion than obsidian or chert. In fact, the rebound hardness range of andesite is wider than those of obsidian and chert (Fig. 4.3.).

On the other hand, chert and obsidian showed higher rebound hardness values. This means that these rocks rebounded most of impact forces, probably because of their homogeneous and microcrystalline structure. Such structures suitable for stone flaking differ from those of stones used for grinding tools, and the differences between them are manifested in the Schmidt Hammer measurements.

In addition, Schmidt Hammer values are likely to be influenced by conditions of raw material nodules. More specifically, during the Schmidt Hammer measurements at the chert outcrops in Jordan, I observed that some nodules of Type FH chert were not suitable for flaking due to many internal flaws that were extensively caused by geological processes like folding and faulting (Fig. 4.8.). Because the geological deposition process like folding and faulting at the outcrops caused the internal flaws. Thus, even if large chert nodules are available, only small parts of them are actually usable for flaking. Nevertheless, the Q-values of Type FH chert nodules are distributed in a higher and narrower range than those od Type M chert.

4.4.2. Rockwell hardness value

The Rockwell indentation caused conchoidal fractures on glass, obsidian, and chart in contrast to metals that deform plastically (Fig. 4.6.). Such fractures occurred as a result of forces that exceeded fracture toughness of glass, chert, and obsidian, and they represent essentially the same phenomena as cracks that occur on chert in the Vickers hardness test (Tsobgou and Dabard, 2010; Schmidt et al., 2019; Moník et al., 2021; Namen et al., 2022). Thus, here I interpret that the Rockwell Hardness values of chert, obsidian, and glass, principally represent the scale of fractures/cracks caused by a pressure from the indenter.

From this viewpoint, the low HRC values of the glass indicate that the large scale of fractures occurred in comparison with the chert and obsidian samples. The increasing order of HRC values from glass to obsidian then chert is consistent with the easiness of flaking I felt in knapping them. Previous studies also suggested that the chert cores required more force to detach flakes than glass and obsidian cores (Dogandžić et al., 2020;

Li et al., 2022).

Based on these observations, it is notable that the HRC values of Type FH chert are significantly lower than Type M because this difference means that the indentation caused larger scale of fracture in Type FH than in Type M. This in turn indicates that Type FH requires less force to fracture. The "knapping force requirement" is considered a significant factor contributing to knapping quality of lithic raw materials (Nickel and Schmidt, 2022). In addition to indentation fracture toughness, Nickel and Schmidt (2022) also take into account tensile strength to estimate the size of a critical crack required to induce stone flaking, i.e., "Griffith" crack length. In their model, a combination of smaller fracture toughness and greater strength leads to smaller length of Griffith cracks, i.e., easiness of flaking. Although this study did not measure tensile strength, rebound hardness measured by Schmidt Hammer has been shown to be positively correlated with compressive strength and elasticity (Young's modulus) (Winkler, 1975; Katz et al., 2000; Yasar and Erdogan, 2004; Aydin and Basu, 2005; Goudie, 2006; Yagiz, 2009; Bilen, 2021; Teymen, 2021). Despite the absence of statistical significance, the results showed an increasing order of rebound hardness from Type M to Type FH then obsidian, which implies greater strength of Type FH than Type M. This observation is principally in accord with the model by Nickel and Schmidt (2022).

4.4.3. Factors of fracture predictability difference between chert types

Why are the appearance of the chert, in this case of surface roughness (R_a value) and translucency, related to the force required for flaking lithics? The previous study suggested that the surface roughness (R_a value) of chert in southern Jordan is positively correlated with the preservation and the abundance of carbonate microfossils such as foraminifers and calcareous algae in chert (Suga et al., 2022). In other words, the Ra value (surface roughness) is increased by better preservation and greater abundance of microfossils. As to the translucency, highly translucent chert contains only a few microfossils that are often poorly preserved and cannot be observed well even under the microscope (Ichinose et al., 2023; Suga et al., 2022). Type FH chert contains few impurities, such as microfossils, and is enriched in SiO₂ (Suga et al., 2022). These properties are expected to make Type FH chert more homogeneous and brittle than Type M, increasing the similarity of Type FH to obsidian and chert. This expectation is consistent with the measurement results of Schmidt Hammer and Rockwell Hardness that indicates greater fracture predictability of Type FH than Type M. However, the relationship between the petrographic features (e.g., microfossils) and mechanical properties is still speculative and requires further studies.

4.4.4. Raw material selectivity of lithic knapping at the Jebel Qalkha area

Based on the above observations about the differences in mechanical properties related to fracture predictability between Type FH and M chert, here I discuss the reasons for the increase in the use of Type FH in the EUP assemblage at the Jebel Qalkha area. As mentioned above, both Type FH and Type M chert are available in local chert outcrops, and I expect that the increase in Type FH chert primarily reflect the changes in raw material selection by prehistoric inhabitants in this area.

As discussed above, I suggest that the knapping force requirement is smaller in Type FH than in Type M. This means that, in the production of flaked stone tools, the strength of percussion can be smaller. Its behavioral implication is that the swing of a hammer can be smaller or slower. A small or slow swing allows a knapper to control the movement of a hammer and thus increase the precision of striking. This should have been particularly beneficial for producing bladelets with tiny platforms in the EUP (i.e., Ahmarian industry). The previous study showed that the tiny platform types such as the linear and punctiform increased in the EUP assemblage in southern Jordan (Kadowaki et al., 2021). In addition, it is widely recognized that the detachment of bladelets with small platforms was achieved by soft-hammer percussion (Ohnuma and Bergman, 1990; Kuhn et al., 2009; Meignen, 2012). This is likely another factor that favored the use of Type FH chert because of its greater fracture predictability that requires less force than Type M. In this way, I suggest that the use of Type FH chert increased because it was more suitable for the production of bladelets that increased in the EUP.

It is also possible that the use of Type FH chert increased in the EUP (Ahmarian) as a result of changes in raw material sources. This is indicated by differences in chemical compositions of some EUP chert artifacts from those of the LMP and IUP in the Jebel Qalkha area (Ichinose et al., 2023). However, the possibility of changes in raw material sources is not exclusive from the intentional selection of Type FH chert that was suitable for the production of bladelets.

In addition, I suggest that the difference in durability of cutting-edge between the chert types influenced the selection of lithic raw materials. More specifically, intentional selection of more durable chert types reduces the consumption of lithic raw materials. Lerner et al. (2007) indicated that wear accrual rates of lithic assemblages were influenced by relative hardness of lithic raw materials. The study of Oldowan stone tools in South Africa suggested that vein quartz predominated Oldowan assemblages because the cutting-edge of vein quartz is more durable than other raw materials (Caruana and Mtshali, 2018). Moreover, Moník and Hadraba (2016) mentioned some possible relations among

fracture predictability/brittleness, edge sharpness and durability of edges. However, durability test has not been conducted yet for the lithic samples from southern Jordan.

However, the above discussion cannot explain why Type FH chert was not used frequently in the LMP and IUP despite the high predictability of fracture. One possible reason is the frequent occurrences of internal flaws in Type FH chert. The presence of internal flaws practically reduces the useful portion of chert for lithic production. Such actual macroscale conditions in the occurrences of Type FH chert may have limited its usefulness for the production of large blanks, such as Levallois flakes and robust blades, in the LMP and IUP. On the other hand, Type M chert does not show as frequent flaws as Type FH. In addition, the nodules and clasts of Type M chert tend to be larger than those of Type FH in the Jebel Qalkha area (Suga et al., 2022). The internal flaws in Type FH may not have been as problematic for the production of small artifacts like bladelets. In fact, some previous studies suggested an advantage of small lithic production that enables the use of raw materials in various sizes (Close, 2002; Vujević et al., 2017). Thus, actual selection of lithic raw material was likely governed by multiple factors including not only fracture predictability but also the presence/absence of flaws, the size of nodules, the size and function of tools to be manufactured, and flaking techniques/methods to be applied. Such multiple factors are likely to have been involved in various complicated uses of lithic raw materials since the Lower Paleolithic (Agam et al., 2022; Finkel et al., 2023).

4.4.5. Relation to lithic miniaturization

The general chert classification of this study (Type M and Type FH) has been widely recognized for lithic assemblages in the Levant (see Chapter 3.3.3.). The previous observations are consistent with the results of this study suggesting that fine-grained chert (Type FH) tend to be preferentially used for flaking miniature lithics (e.g., microliths). To my knowledge, this study is the first case that explains the increase of fine-grained chert from the IUP to EUP (Ahmarian) in the Levant. However, in Europe, it is known that the use of high quality (finer grained) flint increased in the Early Upper Paleolithic in association with the production of blades and bladelets by soft marginal percussion (Moník and Hadraba, 2016; Tomasso and Porraz, 2016). Although the shift to "high quality flint" tends to be taken for granted by archaeologists, it is actually a cultural process that need to be explained. For this purpose, the examination of fracture predictability can be a useful approach as it provides a testable hypothesis through quantitative measurements.

4.5. Summary

This chapter examined how differences in mechanical properties may have influenced the Paleolithic raw material selection. Type FH chert with high fracture predictability increased in tandem with the increase in bladelet production in southern Jordan. To explain lithic miniaturization like bladelets, Pargeter and Shea (2019) suggested a significance of understanding variability in raw material package sizes and quality within specific foraging radii. The examination of fracture predictability can be a useful approach in explaining the selection of lithic raw materials in relation to lithic technology.

Further examinations are necessary to test the validity of these methods. In this study, the chert with high translucency and low surface roughness showed high fracture predictability. In other regions, some chert assemblages are similar to the samples in southern Jordan (e.g., Ghasidian and Heydari-Guran, 2018; Schmidt and Morala, 2018; Kolobova et al. 2021; Aubry et al., 2022), and the measurement of the fracture predictability between different chert types may be applicable to other regions. In addition, it is necessary to examine fracture predictability of other sedimentary rocks and volcanic rocks. The workability of lithic raw material may need to be examined by combining measurement values with impressions of modern knappers. In this way, further studies are necessary in modelling the relationship among key mechanical properties as well as in conducting flaking experiments (e.g., Bradbury et al., 2008; Harmand, 2009).

Quantification of the fracture predictability has an advantage that it can be easily combined with other quantitative lithic data. For example, fracture predictability can be incorporated into multivariate analyses conducted by 3D morphology (e.g., Valletta et al., 2020; Archer et al., 2021; Herzlinger et al., 2021; Radinović and Kajtez, 2021; Muller et al., 2022). Fracture predictability can also be combined with the model that quantifies attractiveness of raw material sources, as mentioned in Introduction (Wilson et al., 2007; Browne and Wilson, 2011). Moreover, its wider applications are expected to facilitate objective comparisons of lithic raw materials between different regions.



Fig. 4.1. a) Schmidt Hammer used in this study. b) Punch marks on obsidian, left by Schmidt Hammer after measurement. c) Scene during the use of Schmidt Hammer on a grinding quern made of andesite.



Fig. 4.2. Samples used for Rockwell Hardness test. A) Slab of Type FH chert. b) Slab of Type M chert. c) Glass block. d) Rockwell Hardness tester during its use for the measurement of a chert sample.



Fig. 4.3. Boxplots showing the Q-values of Type M chert, Type FH chert, Obsidian and Andesite. See Appendix Table S2 for data sources.



Fig. 4.4. Relative frequency histograms of Q-values of Type M chert and Type FH chert. The Kurtosis, that indicates whether distribution is concentrated near average value, is - 0.873 for Type M chert and -0.424 for Type FH chert. See Appendix Table S2 for data sources.



Fig. 4.5. Scatterplots and regression lines of Q-values and mass of chert and obsidian samples. The plots are marked differently according to Type M chert, Type FH chert and Obsidian. See Appendix Table S2 for data sources.



Fig. 4.6. Fractures caused by indentations of Rockwell Hardness test. a–b) plan views of fractures on glass blocks. c–d) profile views of fractures inside glass blocks. e) plan view of fractures on Type FH chert. f) plan view of fractures on Type M chert.



Fig. 4.7. Boxplots showing Rockwell Hardness values of Type M chert, Type FH chert, Obsidian and Glass. See Appendix Table S3 for data sources



Fig. 4.8. Examples of Type FH chert nodules (1-3) and lithics (4-5) with internal flaws

Tuble: The realized of samples of the material types and of measurement methods.						
	Type M	Type FH	Obsidian	Glass	Andesite	
	chert	chert	Obsidiali	Ulass	Allucsic	
Schmidt Hammer	14	13	9		0 7	
Rockwell Hardness (HRC)	6	10	3		1 0	

Table. 4.1. Number of samples by raw material types and by measurement methods.

Table. 4.2. Results of t-tests for pairwise comparisons of rebound hardness.

Compared pair	p-value		
Type FH chert	Type M chert	0.169	
Type FH chert	Obsidian	0.825	
Type FH chert	Andesite	< 0.001	
Type M chert	Obsidian	0.163	
Type M chert	Andesite	< 0.001	
Obsidian	Andesite	< 0.001	

Table. 4.3. Statistical values of linear regressions of mass (kg) vs. Q-value.

Raw material	Linear regression	\mathbb{R}^2	p-value	
Type FH chert	y = 0.09x + 78.56	0.345	0.035	
Type M chert	y = 0.18x + 75.08	0.538	0.003	
Obsidian	y = 0.21x + 78.10	0.251	0.170	

Table. 4.4. Results of Mann-Whitney's U tests of Rockwell Hardness values for pairwise comparisons of raw material types.

Compared pair of	p-value		
Type FH chert	Type M chert	< 0.001	
Type FH chert	Obsidian	< 0.001	
Type M chert	Obsidian	< 0.001	

Chapter 5. Discussion

5.1. Quantitative characterization of lithic raw material variations and fracture predictability

This dissertation is one of the case studies that employ a quantitative approach to examine raw material changes accompanied by lithic miniaturization. Previous studies in the Levant suggested a connection between microlithic production and high knapping quality raw material (fine-grained chert) in the Epipaleolithic (Delage, 2007; Marder and Goring-Morris, 2020). This dissertation suggested that the connection had already appeared in the EUP (Ahmarian). The employment of the quantitative approach allowed this study to show a raw material change in the earlier period than previous studies. In this dissertation, lithic raw material (chert) was classified based on objective attributes (Chapter 3), and the difference of knapping force requirement between the chert types was quantified by the mechanical properties (Chapter 4). On the other hand, some studies explained raw material knapping force requirement from the lithic raw material macroscopic observation, and their explanation is rather similar to a description of their impressions. The quantitative method enabled a precise detection of the timing when raw material change began to appear. A unique contribution of this dissertation is a precise illustration of lithic raw material change through the employment of the quantitative approach.

Another unique feature of this dissertation was to discuss empirically the resource utilization in the Paleolithic through mechanical hardness of lithic raw material. In Chapter 4.1., I described some archaeological case studies that analyzed the fracture predictability mechanically and explained mechanical properties employed in these cases. However, most of these studies were carried out in the context of examining the influence of heat treatment on lithic raw material or examining the differences between high knapping quality raw material (e.g., obsidian) and low knapping quality raw material (e.g., andesite, basalt) recognized commonly among archaeologists. The former examination is based on the hypothesis that the heat treatment increases the predictability of flaking, and a few studies did not consider specific archaeological records. The latter examination is also not empirical because these studies did not analyze archaeological raw materials excavated from the same site. In other words, most of the archaeological studies with mechanical hardness have not considered the actual archaeological record. This dissertation began with an analysis of raw material of lithic assemblage excavated from sites and employed the concept of the fracture predictability in order to interpret the differences in the relative frequencies of the chert types in four lithic assemblages.

Therefore, this dissertation examined fracture predictability using mechanical hardness based on actual archaeological records.

5.2. Changes in the raw material selection at the MP-UP transition

As for the lithic technology of the period covered by this dissertation, the IUP lithic assemblages in the Levant are known for the transitional characteristics between the Middle Paleolithic and the Upper Paleolithic (Meignen, 2012; Goring-Morris and Belfer-Cohen, 2020; Goder-Goldberger and Malinsky-Buller, 2022). Namely, the change from the LMP to the IUP is considered to have been gradual. On the other hand, Kuhn et al. (2009) pointed out the "saltational technological shift" between the IUP and the EUP (Ahmarian), and Goring-Morris and Belfer-Cohen (2018) suggested that it is difficult to derive the lithic industry of the Ahmarian from the IUP assemblage. This dissertation did not detect a transitional shift of lithic raw material between the LMP and the IUP but showed a clear change between the IUP and the EUP. Several studies suggested an influence of raw material knapping quality on the organization of lithic technology (Bamforth, 1986; Andrefsky, 1994, 2009; Kuhn, 2020), thus the suitable resource utilization may have caused an innovation of the lithic technology between the IUP and the EUP.

This dissertation examined lithic raw material changes in southern Jordan of the Levant region in connection with lithic miniaturization. In the previous studies, the phenomena of lithic miniaturization is characterized by the systematic production and utilization of miniature stone tools (Pargeter and Shea, 2019; Shipton, 2023). The lithic miniaturization has long been recognized among archaeologists in various regions and was addressed as a special issue of Archaeological Papers of the American Anthropological Association (Kuhn and Elston, 2002 and papers therein). Recently, a lithic raw material change with lithic miniaturization was also reported in the European Upper Paleolithic (Ahmarian/Aurignacian) (e.g., Moník and Hadraba, 2016; Tomasso and Porraz, 2016) and the Middle to Later Stone Age transition in Africa (Shipton et al., 2018, 2021). Considering these examples, it is possible to hypothesize that the raw material change with lithic miniaturization is one of the Paleolithic adaptive behaviors that can occur in any region, depending on the site and resource conditions (e.g., site location, distance from raw material sources, assumed hunting behavior, etc.). The interpretation is that appropriate resource utilization made miniaturization possible. In order to explain this hypothesis as a general cultural process, it is necessary to increase similar case studies in other regions. In addition, it is also necessary to explain the process and factor that caused the raw material change, for example, why the best lithic raw material was not used at the beginning. As for such factors, this dissertation referred to the difference in the size of cobbles/nodules and frequent occurrences of internal flaws. The actual situation of resource utilization in the Paleolithic can be clarified by analyzing not only the easiness

of flaking but also various aspects of lithic raw material. This dissertation is also one of case studies to generalize the resource utilization pattern with lithic miniaturization and their connection to specific lithic raw material.

5.3. Further research issues

As presented separately in Chapters 3 and 4, this dissertation leaves many further research issues. Here I highlight a few of these issues and present some preliminary analyses and results to overcome these issues.

5.3.1. Frequency of chert types in the Epipaleolithic

One of the remaining issues is a paucity of raw materials data in the EUP lithic assemblage that showed a change in the relative frequencies of chert types. The EUP lithic assemblage in this dissertation is only the assemblage from Layer F–G at Tor Hamar. Accordingly, it is difficult to determine whether the prehistoric people at Tor Hamar were unique or whether the increase of Type FH chert in the EUP is a general phenomena. To examine this issue, I made a preliminary study to see whether the predominant trend of Type FH chert continues in later periods (the Epipaleplithic). Here the assemblage from Layer E2 at Tor Hamar was additionally analyzed. This layer reported radiocarbon dates around 24–18 ka cal. BP (Naito et al., 2022). Previous studies (Kadowaki and Henry, 2019; Naito et al., 2022) expected that the lithic industry from Layer E2 corresponds to "Qalkhan". The Qalkhan, like the Ahmarian of Layer F-G, is also a lithic industry dominated by bladelets (Henry, 1995; Olszewski, 2006). Based on the same method as in Chapter 3, the lithic artifacts were classified by the three chert type-groups, and their relative frequencies were compared with those of the other lithic assemblages in the Jebel Qalkha area.

Fig. 5.1. shows the relative frequencies of the three chert type-groups in the five lithic assemblages. The Epipaleolithic assemblage (Tor Hamar Layer E2) is characterized by a greater proportion of Type FH chert than the EUP assemblage (Tor Hamar Layer F–G). As a result, the dominant use of Type FH chert not only occurred in Layer F–G (Ahmarian) but also in Layer E2 (Qalkhan). This result supports the adaptive behavior of selecting Type FH chert for the production of bladelets, although it is necessary to verify the results with EUP lithic assemblages other than Tor Hamar.

5.3.2. Linking mechanical and structural properties of chert: Loss on ignition

In Chapter 4, the difference in Rockwell hardness between the chert types was statistically significant. However, no other studies have employed Rockwell hardness measurement to examine the fracture predictability, thus it is difficult to assess the result in comparison with other studies. Moreover, this dissertation employs mechanical approaches, but it is necessary to relate the results to the structural approach. Therefore, a preliminary analysis was made to measure the SiO₂ content by XRF and the amount of

impurities within chert samples by Loss on ignition (LOI) in the same method as the previous studies (Williams et al., 2019; Lewis et al., 2022). The impurities within chert samples stem from in situ volatile elements and secondary mineralization (Lewis et al., 2022). Silica, the main component of chert, is not considered a volatile compound. Thus, the SiO₂ content of chert is inversely proportional to the amount of impurities. In previous studies (Williams et al., 2019; Lewis et al., 2022), fracture predictability is represented by the Euclidean distance converting %LOI value and %SiO₂ value. As a result, a scatter plot can illustrate the distance from a theoretically "perfect" chert sample with 100% SiO₂. The previous studies analyzed chert artifacts excavated from the Paleoindian sites based on the above theory and compared the degrees of fracture predictability by chert sources or various periods. This dissertation followed their theory and used Rockwell hardness chert nodule samples (n = 6) for the XRF and LOI measurements. The analyses were conducted by Dr. Koshi Yamamoto (Nagoya University Museum) and Mr. Bayart Nadmid (Graduate School of Environmental Studies, Nagoya University).

As a result, the measurement values of %SiO₂ and %LOI are shown in a scatter plot (Fig. 5.2.). This dissertation did not calculate the Euclidean distance because the purpose is to make simple comparisons and see correspondences among the measurement values. The scatter plot shows that the distribution range of Type FH chert is different from Type M chert. Type FH chert has a higher value of %SiO₂ and is closer to a "perfect" chert than Type M chert. Although a few samples of this dissertation cannot show statistically significant differences, this result is consistent with the previous studies. On the other hand, the measurement values of both %SiO₂ and %LOI are distributed in a narrower range than the previous studies. Thus, this dissertation discussed a slighter difference of the fracture predictability than the previous studies. This result may explain why differences in the Schmidt Hammer hardness were not significant between Type FH and Type M chert.

In addition, Fig. 5.3. shows the relationship between Rockwell hardness values and %SiO₂. Rockwell hardness and the SiO₂ content show a negative correlation. This result suggests that the SiO₂ content of chert influences the knapping force requirement. However, these results have to be verified by additional analyses of more samples.



Fig. 5.1. Diachronic changes in the use of lithic raw material (chert) from the MP to the Early Epipaleolithic (Qalkhan) in southern Jordan.



Fig. 5.2. Scatterplot of the percentages of SiO₂ (silica) and the loss on ignition for each chert sample of Rockwell Hardness (HRC). See Appendix Table S3 for data sources.



Fig. 5.3. Scatterplot showing the correlation between Rockwell Hardness values and the percentage of SiO_2 (silica). Rockwell Hardness values are the average values for each mother nodule. See Appendix Table S3 for data sources.

Chapter 6. Conclusion

This dissertation aimed at diachronic examinations of the selection and utilization of lithic raw material at the MP-UP transition, which is known for the key paleoanthropological processes. Firstly, I analyzed the raw material (chert) variations in the four lithic assemblages that represent the LMP, the IUP and the EUP in the Jebel Qalkha area, southern Jordan. More specifically, three major chert types (Types M, FL and FH) were established based on two macroscopic criteria (texture and translucency) that correlate with microfossil contents and major chemical compositions. As a result, the relative frequencies of the three chert types in the four lithic assemblages showed an increase in Type FH in the EUP assemblage. As for this increase, I also hypothesized that the physical properties of Type FH chert are different from those of the other chert types (Type FL and M). Secondly, I examined the fracture predictability between Type FH and Type M through the measurement of two mechanical properties, i.e., the rebound hardness by Schmidt Hammer and the Rockwell Hardness (HRC), by using chert samples from raw material sources. The results indicate that Type FH chert needs less knapping force requirement and is particularly suitable for the production of bladelets. On the other hand, Type FH chert in southern Jordan suffers from abundant internal fractures and was not used as often as medium-grained chert for Levallois products and robust blades in the LMP and IUP. Depending on end products, specific chert may have been selected at raw material sources where various chert types could be collected. The general discussion presented some implications of this dissertation and several remaining issues. As for the latter, some preliminary analyses and results were presented as further research directions.

This dissertation aimed at contributing to better comprehension of the changes in cultural remains at the MP-UP transition when significant paleoanthropological processes occurred, such as the wide geographic dispersal of *Homo sapiens* and their interaction and interbreeding with Neanderthals and Denisovans. As a result, a clear change in lithic raw material was not found at the conventional MP-UP boundary but detected at a little later phase between the IUP and EUP. However, this result is only for the Jebel Qalkha area. The discussion of behavioral change throughout the MP-UP transition requires investigations in the other region.

At present, the comprehension among archaeologists is still limited regarding the utilization of lithic raw material resource and the mechanical properties of raw material in the Paleolithic. However, this dissertation provided a novel perspective on these issues to make contributions to the further development of the research on prehistoric resource utilization.

Acknowledgements

I sincerely would like to thank Dr. Seiji Kadowaki (my supervisor) for their constructive discussions and guidance on my doctoral study. I wish to thank Dr. Kazuhiro Tsukada and Dr. Hidekazu Yoshida for their constant encouragement and comments. Moreover, I am grateful to all Seiken and Museum Seminar members of Nagoya University for useful discussions. I am deeply grateful to the late Prof. Donald O. Henry for his warm support and encouragement in the renewed investigations of prehistoric site in southern Jordan.

I am grateful to Dr. Atsushi Uemine of Nanzan University for instruction in measuring average surface roughness (R_a). Moreover, I am also grateful to Kazuhiro Kobayashi of the Graduate School of Science/School of Science Equipment Development Support Section, Nagoya University for instruction in measuring Rockwell Hardness (HRC). I am grateful to Dr. Koshi Yamamoto and Bayart Nadmid for their cooperation of the XRF measurement.

I would like to thank Sate Massadeh (Ministry of Tourism and Antiquities of Jordan) and Oday Tarawneh (Department of Antiquities of Jordan) for generous supports of the fieldwork and my study. Moreover, I am grateful for a permission of fieldwork in south Jordan and generous supports from Fadi Balawai (Director General), Aktham Oweidi, and other staff members of the Department of Antiquities of Jordan. I also thank Maher Amreen and other members of the Aqaba Antiquities. The fieldwork in Jordan was accomplished by diligent work by crew members and local communities in the Humayma area.

This research was supported by the Grants-in-Aid for Scientific Research from The Ministry of Education, Culture, Sports, Science and Technology, Japan (grant numbers 16H06409 and 20H00026), JPJSBP (No. 120228403) and JSPS Research Fellows (No. 21J20717/22KJ1518).

References

- Abadi, I., Bar-Yosef, O., Belfer-Cohen, A., 2020. KebaraV-A Contribution for the study of the Middle-Upper Paleolithic Transition in the Levant. *PaleoAnthropology* 2020, 1–28. https://doi.org/10.4207/PA.2020.ART139
- Abbas, M., Lai, Z., Jansen, J. D., Tu, H., Alqudah, M., Xu, X., Al-Saqarat, B., S., Al Hseinat, M., Ou, X., Petraglia, M. D., Carling, P. A., 2023. Human dispersals out of Africa via the Levant. *Science Advances* 9, eadi6838. https://doi.org/10.1126/sciadv.adi6838
- Abrunhosa, A., Pereira, T., Márquez, B., Baquedano, E., Arsuaga, J. L. and Pérez-González, A., 2019. Understanding Neanderthal technological adaptation at Navalmaíllo Rock Shelter (Spain) by measuring lithic raw materials performance variability. *Archaeological and Anthropological Sciences* 11, 5949–5962. https://doi.org/10.1007/s12520-019-00826-3
- Abulafia, T., Goder-Goldberger, M., Berna, F., Barzilai, O., Marder, O., 2021. A technological analysis of the Ahmarian and Levantine Aurignacian assemblages from Manot Cave (area C) and the interrelation with site formation processes. *Journal of Human Evolution* 160, 102707. https://doi.org/10.1016/j.jhevol.2019.102707
- Agam, A., 2020. Late Lower paleolithic lithic procurement and exploitation strategies: A view from Acheulo-Yabrudian Qesem Cave (Israel). *Journal of Archaeological Science: Reports* 33, 102447. https://doi.org/10.1016/j.jasrep.2020.102447
- Agam, A., Wilson, L., 2018. Blind test evaluation of consistency in macroscopic lithic raw material sorting. *Geoarchaeology: An International Journal* 34, 467–477. https://doi.org/10.1002/gea.21720
- Agam, A., Zupancich, A., 2020. Interpreting the Quina and demi-Quina scrapers from Acheulo-Yabrudian Qesem Cave, Israel: Results of raw materials and functional analyses. *Journal of Human Evolution* 144, 102798. https://doi.org/10.1016/j.jhevol.2020.102798
- Agam, A., Rosenberg-Yefet, T., Wilson, L., Shemer, M., Barkai, R., 2022. Flint Type Analysis at Late Acheulian Jaljulia (Israel), and Implications for the Origins of Prepared Core Technologies. *Frontiers in Earth Science* 10, 858032. https://doi.org/10.3389/feart.2022.858032
- Agam, A., Wilson, L., Gopher, A., Barkai, R., 2020. Flint Type Analysis of Bifaces From Acheulo-Yabrudian Qesem Cave (Israel) Suggests an Older Acheulian Origin. *Journal of Paleolithic Archaeology* 3, 719–754. https://doi.org/10.1007/s41982-019-00038-0
- Akazawa, T., Aoki, K., Bar-Yosef, O., 1998. Neanderthals and Modern Humans in Western Asia. Plenum Press, New York. https://doi.org/10.1007/b109961
- Alex, B., Barzilai, O., Hershkovitz, I., Marder, O., Berna, F., Caracuta, V., Abulafia, T., Davis, L., Goder-Goldberger, M., Lavi, R., Mintz, E., Regev, L., Mayer, D., B.-Y.,

Tejero, J.-M., Yeshurun, R., Ayalon, A., Bar-Matthews, M., Yasur, G., Frumkin, A., Latimer, B., Hans, M. G. and Boaretto, E., 2017. Radiocarbon chronology of Manot Cave, Israel and Upper Paleolithic dispersals. *Science Advances* 3, e1701450. https://doi.org/10.1126/sciadv.1701450

- Al-Saqarat, B. S., Abbas, M., Lai, Z., Gong, S., Alkuisi, M. M., Abdalla, M. B., Hamad, A., Carling, P. A. and Jansen, J. D., 2020. A wetland oasis at Wadi Gharandal spanning 125–70 ka on the human migration trail in southern Jordan. *Quaternary Research* 100, 154–169. https://doi:10.1017/qua.2020.82
- Andrefsky, Jr., W., 1994. Raw-material Availability and The Organization of Technology. *American Antiquity* 59, 21–34. https://doi.org/10.2307/3085499
- Andrefsky, Jr., W., 2009. The Analysis of Stone Tool Procurement, Production, and Maintenance. *Journal of Archaeological Research* 17, 65–103. https://doi.org/10.1007/s10814-008-9026-2
- Anikoh, G. A., Adesida, P. A., Afolabi, O. C., 2015. Investigation of Physical and Mechanical Properties of Selected Rock Types in Kogi State Using Hardness Tests. *Journal of Mining World Express* 4, 37. https://doi:10.14355/mwe.2015.04.004
- Archer, W., Djakovic, I., Brenet, M., Bourguignon, L., Presnyakova, D., Schlager, S., Soressi, M., McPherron, S. P., 2021. Quantifying differences in hominin flaking technologies with 3D shape analysis. *Journal of Human Evolution* 150, 102912. https://doi.org/10.1016/j.jhevol.2020.102912
- Assaf, E., 2021. Throughout the generations: Learning processes and knowledge transmission mechanisms as reflected in lithic assemblages of the terminal Lower Paleolithic Levant. *Journal of Archaeological Science: Reports* 35, 102772. https://doi.org/10.1016/j.jasrep.2020.102772
- Aubry, T., Barbosa, A. F., Gameiro, C., Luís, L., Santos, A. T., Silvestre, M., 2022. Far from flint: Inferring land-use and social networks from Middle and Upper Palaeolithic lithic assemblages (Cardina-Salto do Boi, Côa Valley, Portugal). *Journal of Archaeological Science: Reports* 42, 103385. https://doi.org/10.1016/j.jasrep.2022.103385
- Aubry, T., Barbosa, A. F., Luís, L., Santos, A. T., Silvestre, M., 2016. Quartz use in the absense of flint: Middle and Upper Paleolithic raw material economy in the Côa Valley (North-eastern Portugal). *Quaternary International* 424, 113–129. https://doi.org/10.1016/j.quaint.2015.11.067
- Aydin, A., 2009. ISRM Suggested method for determination of the Schmidt hammer rebound hardness: Revised version. *International Journal of Rock Mechanics & Mining Sciences* 46, 627–634. https://doi.org/10.1016/j.ijrmms.2008.01.020
- Aydin, A., Basu, A., 2005. The Schmidt hammer in rock material characterization. *Engineering Geology* 81, 1–14. https://doi.org/10.1016/j.enggeo.2005.06.006
- Bae, C. J., Douka, K., Petragila, M. D., 2017. On the origin of modern humans: Asian perspectives. *Science* 358, eaai9067. https://doi.org/10.1126/science.aai9067
- Bamforth, D. B., 1986. Technological Efficiency and Tool Curation. American Antiquity

51, 38–50. https://doi.org/10.2307/280392

- Bamforth, D. B., 2009. Projectile points, people, and Plains Paleoindian perambulations. *Journal of Anthropological Archaeology* 28, 142–157. https://doi.org/10.1016/j.jaa.2009.01.002
- Banning, E. B., 2020. The Archaeologist's Laboratory: The Analysis of Archaeological Evidence, Second Edition. Springer, Cham. https://doi.org/10.1007/978-3-030-47992-3
- Bar-Yosef, O., 2000. The Middle and Early Upper Paleolithic in southwest Asia and neighboring regions. In: Bar-Yosef, O., Pilbeam, D. (Eds.), *The geography of Neandertals and modern humans in Europe and the Greater Mediterranean*. Harvard University, Cambridge, pp. 107–156.
- Bar-Yosef, O., Belfer-Cohen, A., 2019. The Upper Paleolithic industries of Kebara Cave. In: Meignen, L., Bar-Yosef, O. (Eds.), *Kebara Cave, Mt. Carmel, Israel. The Middle* and Upper Paleolithic Archaeology. Part II. Peabody Museum Press, Cambridge, pp. 309–401.
- Barzilai, O., 2022. The origins and destinations of the Levantine Initial Upper Paleolithic – a view from the Negev Desert, Israel. Acta Anthropologica Sinica 41. https://doi.org/10.16359/j.1000-3193/AAS.2022.0035.
- Baykara, İ., Mentzer, S. M., Stiner, M. C., Asmerom, Y., Güleç E. S., Kuhn, S. L., 2015. The Middle Paleolithic occupations of Üçağızlı II Cave (Hatay, Turkey): Geoarcheological and archeological perspectives. *Journal of Archaeological Science: Reports* 4, 409–426. http://dx.doi.org/10.1016/j.jasrep.2015.09.022
- Belfer-Cohen, A., Goring-Morris, N., 2018. An Anthropological Review of the Upper Paleolithic in the Southern Levant. In: Yasur-Laudau, A., Cline, E. H., Rowan, Y. M. (Eds.), *The Social Archaeology of the Levant from prehistory to the present*. Cambridge University Press, New York, pp. 29–46. https://doi.org/10.1017/9781316661468.004
- Belfer-Cohen, A., Hovers. E., 2010. Modernity, Enhanced Working Memory, and the Middle to Upper Paleolithic Record in the Levant. *Current Anthropology* 51, 167– 175. https://doi.org/10.1086/649835
- Bellar, J. A., 2023. Raw material characterization and lithic provenance from Shishan Marsh 1, a Middle Pleistocene site in the central Azraq Basin, Jordan. *Journal of Archaeological Science: Reports* 47, 103800. https://doi.org/10.1016/j.jasrep.2022.103800
- Beller, J. A., Ames, C. J. H., Nowell, A., 2020. Exploring Mid-Late Pleistocene lithic procurement strategies at Shishan Marsh 1: Preliminary geochemical characterization of chert sources around the Greater Azraq Oasis Area, Jordan. *Journal of Archaeological Science: Reports* 29, 102091. https://doi.org/10.1016/j.jasrep.2019.102091
- Betts, M. W., Latta, M. A., 2000. Rock Surface Hardness as An Indication of Exposure Age: An Archaeological Application of The Schmidt Hammer. *Archaeometry* 42, 209–223. https://doi.org/10.1111/j.1475-4754.2000.tb00877.x
- Bhushan, B., 2001. Surface Roughness Analysis and Measurement Techniques. In:
Bhushan, B. (Ed.), *Modern tribology handbook volume 1 Principles of tribology*. CRC Press, Boca Raton, pp. 49–120. https://doi.org/10.1201/9780849377877

- Bilen, C., 2021. Schmidt Hammer Rebound (SHR) values as a guide for the initial estimates of limestone uniaxial compressive strength (UCS) and Hardgrove grindability index (HGI). *Arabian Journal of Geosciences* 14, 2184. https://doi.org/10.1007/s12517-021-08421-4
- Binford, L. R., 1979. Organization and Formation Processes: Looking at Curated Technologies. *Journal of Anthropological Research* 35, 255–273. https://doi.org/10.1086/jar.35.3.3629902
- Boaretto, E., Barkai, R., Gopher, A., Berna, F., Kubik, P. W., Weiner, S., 2009. Specialized Flint Procurement Strategies for Hand Axes, Scrapers and Blades in the Late Lower Paleolithic: A ¹⁰Be Study at Qesem Cave, Israel. *Human Evolution* 24, 1–12.
- Boaretto, E., Hernandez, M., Gober-Goldberger, M., Aldeias, V., Regev, L., Caracuta, V., McPherron, S. C., Hublin, J.-J., Weiner, S., Barzilai, O., 2021. The absolute chronology of Boker Tachtit (Israel) and implications for the Middle to Upper Paleolithic transition in the Levant. *Proceedings of the National Academy of Sciences of the United States of America* 118, e2014657118. https://doi.org/10.1073/pnas.2014657118
- Boëda, E., Bonilauri, S., 2006. The Intermediate Paleolithic: The First Bladelet Production 40,000 Years Ago. *Anthropologie* 44, 75–92.
- Bosch, M. D., Mannino, M. A., Prendergast, A. L., O'Connell, T. C., Demarchi, B., Taylor, S. M., Niven, L., van der Plicht, J., Hublin, J.-J., 2015. New chronology for Ksâr 'Akil (Lebanon) supports Levantine route of modern human dispersal into Europe. *Proceedings of the National Academy of Sciences of the United States of America* 112, 7683–7688. https://doi.org/10.1073/pnas.1501529112
- Bradbury, A., Carr, P., Cooper, D., 2008. Raw Material And Retouched Flakes. In: Andrefsky, Jr. W. (Ed.), *Lithic Technology: Measures of Production, Use and Curation*. Cambridge University Press, Cambridge, pp. 233–254. https://doi.org/10.1017/CBO9780511499661.012
- Brandl, M., Martinez, M. M., Hauzenberger, C., Filzmoser, P., Nymoen, P., Mehler, N., 2018. A multi-technique analytical approach to sourcing Scandinavian flint: Provenance of ballast flint from the shipwreck "Leirvigen 1", Norway. *PLoS ONE* 13(8), e0200647. https://doi.org/10.1371/journal.pone.0200647
- Brantingham, P. J., Olsen, J. W., Rech, J. A., Krivoshapkin, A. I., 2000. Raw Material Quality and Prepared Core Technologies in Northeast Asia. *Journal of Archaeological Science* 27, 255–271. https://doi.org/10.1006/jasc.1999.0456
- Braun, D. R., Plummer, T., Ferraro, J. V., Ditchfield, P., Bishop, L. C., 2009. Raw material quality and Oldowan hominin toolstone preferences: evidence from Kanjera South, Kenya. *Journal of Archaeological Science* 36, 1605–1614. https://doi.org/10.1016/j.jas.2009.03.025
- Broitman, E., 2017. Indentation Hardness Measurements at Macro-, Micro-, and Nanoscale: A Critical Overview. *Tribology Letters* 65, 23. https://doi.org/10.1007/s11249-016-0805-5

- Brown, E. T. (Ed.), 1981. *Rock characterization testing and monitoring: ISRM suggested methods*. Pergamon Press, Oxford.
- Brown, K. S., Marean, C. W., Herries, A. I. R., Jacobs, Z., Tribolo, C., Braun, D., Roberts, D. L., Meyer, M. C., Bernatchez, J., 2009. Fire As an Engineering Tool of Early Modern Humans. *Science* 325, 859–862. https://doi.org/10.1126/science.1175028
- Browne, C. L., Wilson, L., 2011. Resource selection of lithic raw materials in the Middle Paleolithic in southern France. *Journal of Human Evolution* 61, 597–608. https://doi.org/10.1016/j.jhevol.2011.08.004
- Bustillo, M. A., Castañeda, N., Capote, M., Consuegra, S., Criado, C., Díaz-del-Río, P. Orozco, T., Pérez-Jiménez, J. L., Terradas, X., 2009. Is the macroscopic classification of flint useful? A petroarchaeological analysis and characterization of flint raw materials from the Iberian Neolithic mine of Casa Montero. *Archaeometry* 51, 175–196. https://doi.org/10.1111/j.1475-4754.2008.00403.x
- Callahan, E., 1979. The Basics Of Biface Knapping In The Eastern Fluted Point Transition: A Manual For Flintknappers And Lithic Analysts. *Archaeology of Eastern North America* 7, 1–180.
- Carvalho, M., Bicho, N., 2022. Complexity in the Middle to Upper Paleolithic Transition in Peninsular Southern Europe and application of refugium concepts. *Journal of Quaternary Science* 37, 380–393. https://doi.org/10.1002/jqs.3350
- Caruana, M. V., Mtshalib, S. P., 2018. Assessing the Durability of Oldowan Stone Tools in South Africa: Implications for Quartz Selectivity and Use. *Lithic Technology* 43, 245–254. https://doi.org/10.1080/01977261.2018.1516721
- Caux, S., Galland, A., Queffelec, A., Bordes, J., 2018. Aspects and characterization of chert alteration in an archaeological context: A qualitative to quantitative pilot study. *Journal of Archaeological Science: Reports* 20, 210–219. https://doi.org/10.1016/j.jasrep.2018.04.027
- Clark, G. A., Coinman, N. R., Hill, J. B., Neeley, M. P., Olszewski, D. I., Peterson, J. D., Schuldenrein, J., 2017. Survey and Excavation of Stone Age Sites in Jordan's Wadi al-Hasa: 1979–2012. In: Enzel, Y. and Bar-Yosef, O. (Eds.), *Quarternary of the Levant: Environments, Climate Change, and Humans*. Cambridge University Press, New York, pp. 315–328. https://doi.org/10.1017/9781316106754.037
- Close, A. E., 2002. Backed Bladelets Are a Foreign Country. Archaeological Papers of the American Anthropological Association 12, 31–44. https://doi.org/10.1525/ap3a.2002.12.1.31
- Cook, R.F., Pharr, G.M., 1990. Direct Observation and Analysis of Indentation Cracking in Glasses and Ceramics. *Journal of the American Ceramic Society* 73, 787–817. https://doi.org/10.1111/j.1151-2916.1990.tb05119.x
- Copeland, L., 1975. The Middle and Upper Paleolithic of Lebanon and Syria in the Light of Recent Research. In: Wendorf, F., Marks, A. E. (Eds.), *Problem in Prehistory: North Africa and the Levant.* SMU Press, Drass, pp. 317–350.
- Copeland, L., 2000. Forty-six Emireh Points from The Lebanon in The Context of The Middle to Upper Paleolithic Transition in The Levant. *Paléorient* 26, 73–92. https://doi.org/10.3406/paleo.2000.4700

- Cotterell, B., Kamminga, J., 1987. The Formation of Flakes. *American Antiquity* 52, 675–708. https://doi.org/10.2307/281378
- Cotterell, B., Kamminga, J., Dickson, F. P., 1985. The essential mechanics of conchoidal flaking. *International Journal of Fracture* 29, 205–221. https://doi.org/10.1007/BF00125471
- Delage, C., 2003. Siliceous Rocks And Prehistory: Bibliography On Geo-Archaeological Approaches To Chert Sourcing And Prehistoric Explanation. John and Erica Hedges, Oxford.
- Delage, C., 2007. Chert Availability and Prehistoric Exploitation in the Near East: An Introduction. In: Delage, C. (Ed.), *Chert Availability and Prehistoric Exploitation in the Near East.* John and Erica Hedges, Oxford, pp. 1–17.
- Delage, C., Webb, J., 2020. Cherts and prehistory in the Near East... recent advances: An introduction. *Journal of Archaeological Science: Reports* 32, 102446. https://doi.org/10.1016/j.jasrep.2020.102446
- Demidenko, Y. E., Usik, V. I., 2003. Into the Mind of the Maker: Refitting Study and Technological Reconstructions. In: Henry, D. O. (Ed.), Neanderthal in the Levant: Behavioral Organization and the Beginnings of Human Modernity. Continuum, London, pp. 107–155.
- Demirdag, S., Yavuz, H., Altindag, R., 2009. The effect of sample size on Schmidt rebound hardness value of rocks. *International Journal of Rock Mechanics & Mining Sciences* 46, 725–730. https://doi.org/10.1016/j.ijrmms.2008.09.004
- Dennell, R., 2020. From Arabia to the Pacific: How Our Species Colonised Asia. Routledge, London and New York. https://doi.org/10.4324/9781003038788
- Doelman, T., Webb, J., Domanski, M., 2001. Source to discard: patterns of lithic raw material procurement and use in Sturt National Park, northwestern New South Wales. *Archaeology in Oceania* 36, 15–33. https://doi.org/10.1002/j.1834-4453.2001.tb00468.x
- Dogandžić, T., Abdolazadeh, A., Leader, G., Li, L., McPherron, S.P., Tennie, C., Dibble, H. L., 2020. The results of lithic experiments performed on glass cores are applicable to other raw materials. *Archaeological and Anthropological Sciences* 12, 44. https://doi.org/10.1007/s12520-019-00963-9
- Domanski, M., Webb, J., 2007. A Review of Heat Treatment Research. *Lithic Technology* 32, 153–194. https://doi.org/10.1080/01977261.2007.11721052
- Domanski, M., Webb, J.A., 1992. Effect of Heat Treatment on Siliceous Rocks Used in Prehistoric Lithic Technology. *Journal of Archaeological Science* 19, 601–614. https://doi.org/10.1016/0305-4403(92)90031-W
- Domanski, M., Webb, J. A., Boland, J., 1994. Mechanical Properties of Stone Artifact Materials and The Effect of Heat Treatment. *Archaeometry* 36, 177–208. https://doi.org/10.1111/j.1475-4754.1994.tb00963.x
- Domanski, M., Webb, J., Glaisher, R., Gurba, J., Libera, J., Zakoscielna, A., 2009. Heat treatment of Polish flints. *Journal of Archaeological Science* 36, 1400–1408. https://doi.org/10.1016/j.jas.2009.02.002
- Douka, K., 2013. Exploring "the great wilderness of prehistory": The Chronology of the

Middle to the Upper Paleolithic Transition in the Northern Levant. *Mitteilungen der Gesellschaft für Urgeschichte* 22, 11–40.

- Egeland, C. P., Fadem, C. M., Byerly, R. M., Henderson, C., Fitzgerald, C., Mabulla, A.
 Z. P., Baquedano, E., Gidna, A., 2019. Geochemical and physical characterization of lithic raw materials in the Olduvai Basin, Tanzania. *Quaternary International* 526, 99–115. https://doi.org/10.1016/j.quaint.2019.09.036
- Ekshtain, R., Tryon, C. A., 2019. Lithic raw material acquisition and use by early Homo sapiens at Skhul, Israel. *Journal of Human Evolution* 127, 149–170. https://doi.org/10.1016/j.jhevol.2018.10.005
- Ekshtain, R., Zaidner, Y., 2022. Raw material exploitation at the Middle Paleolithic site of Nesher Ramla, Israel. *Quaternary International* 624, 34–48. https://doi.org/10.1016/j.quaint.2021.02.038
- Ekshtain, R., Ilani, S., Segal, I., Hovers, E., 2017. Local and Nonlocal Procurement of Raw Material in Amud Cave, Israel: The Complex Mobility of Late Middle Paleolithic Groups. *Geoarchaeology: An International Journal* 32, 189–214. https://doi.org/10.1002/gea.21585
- Ekshtain, R., Malinsky-Buller, A., Ilani, S., Segal, I., Hovers, E., 2014. Raw material exploitation around the Middle Paleolithic site of 'Ein Qashish. *Quaternary International* 331, 248–266. https://doi.org/10.1016/j.quaint.2013.07.025
- Eren, M. I., Lycett, S. J., Roos, C. I., Sampson, C. G., 2011. Toolstone constraints on knapping skill: Levallois reduction with two different raw materials. *Journal of Archaeological Science* 38, 2731–2739. https://doi.org/10.1016/j.jas.2011.06.011
- Eren, M. I., Roos, C. I., Story, B. A., Cramon-Taubadel, N., Lycett, S. J., 2014. The role of raw material differences in stone tool shape variation: an experimental assessment. *Journal of Archaeological Science* 49, 472–487. https://doi.org/10.1016/j.jas.2014.05.034
- Féblot-Augustins, J., 2009. Revisiting European Upper Paleolithic Raw Material Transfers: The Demise of the Cultural Ecological Paradigm? In: Adams, B., Blades, B. (Eds.), *Lithic Materials and Paleolithic Societies*. Wiley-Blackwell, Oxford, pp. 25–46. https://doi.org/10.1002/9781444311976.ch3
- Finkel, M., Bar, O., Ben Dor, Y., Ben-Yosef, E., Tirosh, O., Sharon, G., 2023. Evidence for sophisticated raw material procurement strategies during the Lower Paleolithic— Hula Valley case study. *Geoarchaeology: An International Journal*, 1–16. https://doi.org/10.1002/gea.21968
- Finkel, M., Barkai, R., Gopher, A., Tirosh, O., Ben-Yosef, E., 2019. The "Flint Depot" of prehistoric northern Israel: Comprehensive geochemical analysis of flint extraction and reduction complexes and implications for provenance studies. *Geoarchaeology: An International Journal* 34, 661–683. https://doi.org/10.1002/gea.21727
- Finkel, M., Erel, Y., Ben Dor, Y., Tirosh, O., Levy, T. E., Najjar, M., Avni, Y., Gopher A., Ben-Yosef, E., 2022. High resolution elemental characterization of prehistoric flint sources in southern Israel: Implications for archaeological provenance studies. *Journal of Archaeological Science: Reports* 43, 103438. https://doi.org/10.1016/j.jasrep.2022.103438

- Finkel, M., Gopher, A., Agam, A., 2020b. Excavating tailing piles at Kakal Spur (Kerem Ben Zimra) locality in the Nahal Dishon prehistoric flint extraction and reduction complex, northern Galilee, Israel. *Archaeological Research in Asia* 23, 100207. https://doi.org/10.1016/j.ara.2020.100207
- Finkel, M., Gopher, A., Barkai, R., 2016. Extensive Paleolithic Flint Extraction and Reduction Complexes in the Nahal Dishon Central Basin, Upper Galilee, Israel. *Journal of World Prehistory* 29, 217–266. https://doi.org/10.1007/s10963-016-9097-9
- Finkel, M., Gopher, A., Ben-Yosef, E., Barkai, R., 2018. A middle paleolithic and neolithic/chalcolithic flint extraction and reduction complex at Mt. Achbara, Eastern Galilee, Israel. Archaeological Research in Asia 16, 14–33. https://doi.org/10.1016/j.ara.2018.01.004
- Finkel, M., Gopher, A., Sharon, G., Tirosh, O., Ben-Yosef, E., 2020a. Kela A Middle Paleolithic flint extraction and reduction site in the northern Golan Heights. *Journal* of Archaeological Science: Reports 30, 102208. https://doi.org/10.1016/j.jasrep.2020.102208
- Fox, J. R., Coinman, N. R., 2004. Emergence of the Levantine Upper Paleolithic. In: Brantingham, P. J., Kuhn, S. L., Kerry, K. W. (Eds.), *The Early Upper Paleolithic beyond Western Europe*. University of California Press, California, pp. 97–112.
- Frahm, E., Hauck, T. C., 2017. Origin of an obsidian scraper at Yabroud Rockshelter II(Syria): Implications for Near Eastern social networks in the early Upper Paleolithic. *Journal of Archaeological Science: Reports* 13, 415–427. https://doi.org/10.1016/j.jasrep.2017.04.021
- Frahm, E., Tryon, C. A., 2019. Origin of an Early Upper Palaeolithic obsidian burin at Ksar Akil (Lebanon): Evidence of increased connectivity ahead of the Levantine Aurignacian? *Journal of Archaeological Science: Reports* 28, 102060. https://doi.org/10.1016/j.jasrep.2019.102060
- Garrod D., 1951. A Transitional Industry from the Base of the Upper Paleolithic in Palestine and Syria. *Journal of the Royal Anthropological Institute* 80, 121–130. https://doi.org/10.2307/2844019
- Garrod, D., 1955. Mugharet el-Emireh in the lower Galilee: Type-Station of the Emiran Industry. *Journal of the Royal Anthropological Institute* 85, 141–162. https://doi.org/10.2307/2844188
- Garvey, R., 2015. A Model of Lithic Raw Material Procurement. In: Goodale, N., Andrefsky, Jr. W. (Eds.), *Lithic Technological Systems and Evolutionary Theory*. Cambridge University Press, Cambridge, pp. 156–171. https://doi.org/10.1017/CBO9781139207775.013
- Gennai, J., Schemmel, M., Richter, J., 2023. Pointing to the Ahmarian. Lithic Technology and the El-Wad Points of Al-Ansab 1. *Journal of Paleolithic Archaeology* 6, 6. https://doi.org/10.1007/s41982-022-00131-x
- Ghasidian, E., Heydari-Guran, S., 2018. Upper Palaeolithic Raw Material Economy in the Southern Zagros Mountains of Iran. In: Nishiaki, Y., Akazawa, T. (Eds.), *The Middle and Upper Paleolithic Archaeology of the Levant and Beyond*. Springer,

Singapore. pp. 157–174. https://doi.org/10.1007/978-981-10-6826-3 11

- Ghorbani, S., Hoseinie, S. H., Ghasemi, E., Sherizadeh, T., 2022. A review on rock hardness testing methods and their applications in rock engineering. *Arabian Journal of Geosciences* 15, 1067. https://doi.org/10.1007/s12517-022-10314-z
- Gilead, I., 1983. Upper Palaeolithic Occurrences in Sinai and the transition to the Epipalaeolithic in the Southern Levant. *Paléorient* 9, 39–53. https://doi.org/10.3406/paleo.1983.4330
- Gilead, I., 1991. The Upper Paleolithic Period in the Levant. *Journal of World Prehistory* 5, 105–154. https://doi.org/10.1007/BF00974677
- Gilead, I., Bar-Yosef, O., 1993. Early Upper Paleolithic Sites in the Qadesh Barnea Area, NE Sinai. Journal of Field Archaeology 20, 265–280. https://doi.org/10.1179/009346993791549200
- Goder-Goldberger, M., 2020. The Middle to Upper Palaeolithic transition as seen from Far'ah II and Boker Tachtit, Israel: Does it relate to the Nile Valley? In: Leplongeon, A., Goder-Goldberger, M., Pleurdeau, D. (Eds.), Not Just a Corridor: Human occupation of the Nile Valley and neighbouring regions between 75,000 and 15,000 years ago. Muséum national d'Histoire naturelle, Paris, pp. 221–237.
- Goder-Goldberger, M., Bar-Matthews, M., 2019. Novel chrono-cultural constraints for the Middle Paleolithic site of Rosh Ein Mor (D15), Israel. *Journal of Archaeological Science: Reports* 24, 102–114. https://doi.org/10.1016/j.jasrep.2018.12.021
- Goder-Goldberger, M., Barzilai, O., Boaretto, E., 2023. Innovative Technological Practices and their Role in the Emergence of Initial Upper Paleolithic Technologies: A View from Boker Tachtit. *Journal of Paleolithic Archaeology* 6, 11. https://doi.org/10.1007/s41982-023-00137-z
- Goder-Goldberger, M., Crouvi, O., Caracuta, V., Horwitz, L. K., Neumann, F. H., Porat, N., Scott, L., Shavit, R., Jacoby-Glass, Y., Zilberman, T., Boaretto, E., 2020. The Middle to Upper Paleolithic Transition in the southern Levant: New insights from the late Middle Paleolithic site of Far'ahII, Israel. *Quaternary Science Reviews* 237, 106304. https://doi.org/10.1016/j.quascirev.2020.106304
- Goder-Goldberger, M., Malinsky-Buller, A., 2022. The Initial Upper Paleolithic and Its Place Within the Middle-to-Upper Paleolithic Transition of Southwest Asia: What Hides Behind the Curtain of Taxonomies? *Journal of Paleolithic Archaeology* 5, 2. https://doi.org/10.1007/s41982-022-00112-0
- Gökkaya, H., Nalbant, M., 2007. The effects of cutting tool geometry and processing parameters on the surface roughness of AISI 1030 steel. *Materials and Design* 28, 717–721. https://doi.org/10.1016/j.matdes.2005.09.013
- Goldman-Neuman, T., Hovers, E., 2012. Raw material selectivity in Late Pliocene Oldowan sites in the Makaamitalu Basin, Hadar, Ethiopia. *Journal of Human Evolution* 62, 353–366. https://doi.org/10.1016/j.jhevol.2011.05.006
- Gómez de Soler, B., Soto, M., Vallverdú, J., Bargalló, A., Chacón, M.G., Romagnoli, F., Vaquero, M., 2020b. The Panadella chert (Montmaneu Formation): a high-quality raw material in the Abric Romaní sequence (NE Iberian Peninsula). Archaeological and Anthropological Sciences 12, 252. https://doi.org/10.1007/s12520-020-01198-9

- Gómez de Soler, B., Soto, M., Vallverdú, J., Vaquero, M., Bargalló, A. Chacón M. G., Romagnoli, F., Carbonell, E., 2020a. Neanderthal lithic procurement and mobility patterns through a multi-level study in the Abric Romaní site (Capellades, Spain). *Quaternary* Science Reviews 237, 106315. https://doi.org/10.1016/j.quascirev.2020.106315
- Goodie, A. S., 2006. The Schmidt Hammer in geomorphological research. *Progress in Physical Geography* 30, 703–718. https://doi.org/10.1177/0309133306071954
- Goodman, M. E., 1944. The Physical Properties of Stone Tool Materials. *American Antiquity* 9, 415–433. https://doi.org/10.2307/275093
- Goring-Morris, N., Belfer-Cohen, A., 2018. The Ahmarian in the Context of the Earlier Upper Paleolithic in the Near East. In: Nishiaki, Y., Akazawa, T. (Eds.), *The Middle* and Upper Paleolithic Archaeology of the Levant and Beyond. Springer, Singapore, pp. 87–104. https://doi.org/10.1007/978-981-10-6826-3_7
- Goring-Morris, N., Belfer-Cohen, A., 2020. Noisy beginnings: The Initial Upper Paleolithic in Southwest Asia. *Quaternary International* 551, 40–46. https://doi.org/10.1016/j.quaint.2020.01.017
- Goring-Morris, N., Davidzon, A., 2006. Straight To The Point: Upper Paleolithic Ahmarian Lithic Technology In The Levant. *Anthropologie* 44, 93–111.
- Grimaldi, S., Porraz, G., Santaniello, F., 2014. Raw material procurement and land use in the northern Mediterranean Arc: insight from the first Proto-Aurignacian of Riparo Mochi (Balzi Rossi, Italy). Quartär 61, 113–127. https://doi.org/10.7485/QU61_06
- Groucutt, H. S., 2014. Middle Paleolithic point technology, with a focus on the site of Tor Faraj (Jordan, MIS3). *Quaternary International* 350, 205–226. https://doi.org/10.1016/j.quaint.2014.06.025
- Groucutt, H. S., Scerri, E. M. L., Amor, K., Shipton, C., Jennings, R. P., Parton, A., Clark-Balzan, L., Alsharekh, A., Petraglia, M. D., 2017. Middle Paleolithic raw material procurement and early stage reduction at Jubbah, Saudi Arabia. *Archaeological Research in Asia* 9, 44–62. https://doi.org/10.1016/j.ara.2017.01.003
- Groucutt, H. S., Scerri, E. M. L., Lewis, L., Clark-Balzan, L., Blinkhorn, J., Jennings, R. P., Parton, A., Petraglia, M. D., 2015. Stone tool assemblages and models for the dispersal of Homo sapiens out of Africa. *Quaternary International* 382, 8–30. https://doi.org/10.1016/j.quaint.2015.01.039
- Harmand, S., 2009. Variability in Raw Material Selectivity at the Late Pliocene sites of Lokalalei, West Turkana, Kenya. In: Hovers, E., Braun, D. R. (Eds.), *Interdisciplinary Approaches to the Oldowan*. Springer, Dordrecht, pp. 85–98. https://doi.org/10.1007/978-1-4020-9060-8_8
- Hauck, T. C., 2011. Mousterian technology and settlement dynamics in the site of Hummal (Syria). Journal of Human Evolution 61, 519–537. https://doi.org/10.1016/j.jhevol.2011.01.014
- Henry, D. O., 1989. From Foraging to Agriculture: The Levant at the End of the Ice Age. University of Pennsylvania Press, Philadelphia.
- Henry, D. O., 1994. Prehistoric cultural ecology in southern Jordan. *Science* 265, 336–341. https://doi.org/10.1126/science.265.5170.336

- Henry, D. O., 1995. Prehistoric Cultural Ecology and Evolution Insights from Southern Jordan. Planum Press, New York. https://doi.org/10.1007/978-1-4757-2397-7
- Henry, D. O., 2011. Late Levantine Mousterian Spatial Patterns At Landscape and Intrasite In Southern Jordan. In: Tensorer, L. J.-M., Jagher, R., Otte, M. (Eds.), *The Lower and Middle Palaeolithic of the Middle East and Neighbouring Regions*. Université de Liège, Liège, pp. 115–129.
- Henry, D. O., 2012. The palimpsest problem, hearth pattern analysis, and Middle Paleolithic site structure. *Quaternary International* 247, 246–266. https://doi.org/10.1016/j.quaint.2010.10.013
- Henry, D. O., 2017a. The Middle Palaeolithic of Southern Jordan. In: Enzel, Y., Bar-Yosef, O. (Eds.), *Quarternary of the Levant: Environments, Climate Change, and Humans*. Cambridge University Press, New York, pp. 585–592. https://doi.org/10.1017/9781316106754.064
- Henry, D. O., 2017b. The Upper and Epipalaeolithic of Southern Jordan. In: Enzel, Y., Bar-Yosef, O. (Eds.), *Quarternary of the Levant: Environments, Climate Change,* and Humans. Cambridge University Press, New York, pp. 659–668. https://doi.org/10.1017/9781316106754.073
- Henry, D. O., Mraz, V., 2020. Lithic economy and prehistoric human behavioral ecology viewed from southern Jordan. *Journal of Archaeological Science: Reports* 29, 102089. https://doi.org/10.1016/j.jasrep.2019.102089
- Henry, D. O., Belmaker, M., Bergin, S. M., 2017. The effect of terrain on Neanderthal ecology in the Levant. *Quaternary International* 435, 94–105. https://doi.org/10.1016/j.quaint.2015.10.023
- Henry, D. O., Nowell, A., Mendez, K., Peterson, E., Senn, M., Rockwell, H., 2014. The Technotypology of Chipped Stone Artifacts. In: Henry, D. O., Beaver, J. E. (Eds.), *The Sands of Time: The Desert Neolithic Settlement at Ayn Abū Nukhayla*. ex oriente, Berlin, pp. 145–170.
- Henry, D.O. (Ed.), 2003. Neanderthals in the Levant: Behavioral Organization and the Beginnings of Human Modernity. Continuum, London.
- Herzlinger, G., Brenet, M., Varanda, A., Deschamps, M., Goren-Inbar, N., 2021. Revisiting the Acheulian Large Cutting Tools of 'Ubeidiya, Israel. *Journal of Paleolithic Archaeology* 4, 31. https://doi.org/10.1007/s41982-021-00108-2
- Higham, T., Douka, K., Wood, R., Ramsey, C. B., Brock, F., Basell, L., Camps, M., Arrizabalaga, A., Baena, J., Barroso-Ruiz, C., Bergman, C., Boitard, C., Boscato, P., Caparros, M., Conard, N. J., Drally, C., Froment, A., Galvan, B., Gambassini, P., Garcia-Moreno, A., Grimaldi, S., Haesaerts, P., Holt, B., Iriarte-Chiapusso, M., Jelinek, A., Jorda Pardo, J. F., Maillo-Fernandez, J., Marom, A., Maroto, J., Menendez, M., Metz, L., Morin, E., Moroni, A., Negrino, F., Panagopoulou, E., Peresani, M., Pirson, S., Rasilla, M., Riel-Salvatore, J., Ronchitelli, A., Santamaria, D., Semal, P., Slimak, L., Soler, J., Soler, N., Villaluenga, A., Pinhasi, R., Jacobi, R., 2014. The timing and spatiotemporal patterning of Neanderthal disappearance. *Nature* 512, 707–717. https://doi.org/10.1038/nature13621

Hirose, M., Belmaker, M., Kadowaki, S., Massadeh, S., Henry, D. O., 2022.

Epipaleolithic Hunting in an Arid Area of the Levant: Faunal Remains from Tor Hamar, Southern Jordan. *ORIENT* 57, 21–42.

- Holt, B., Negrino, F., Riel-Salvatore, J., Formicola, V., Arellano, A., Arobba, D., Boschian, G., Churchill, S. E., Cristiani, E., Canzii, E. D., Vicino, G., 2019. The Middle-Upper Paleolithic transition in Northwest Italy: new evidence from Riparo Bombrini (Balzi Rossi, Liguria, Italy). *Quaternary International* 508, 142–152. https://doi.org/10.1016/j.quaint.2018.11.032
- Hovers, E., 2009. *The lithic assemblages of Qafzeh Cave*. Oxford University Press, New York. https://doi.org/10.1093/oso/9780195322774.001.0001
- Hublin, J.-J., Sirakov, N., Aldeias, V., Bailey, S., Bard, E., Delvigne, V., Endarova, E., Fagault, Y., Fewless, H., Haidinjak, M., Kromer, B., Krumov, I., Marreiros, J., Martisius, N. L., Paskulin, L., Sinet-Mathiot, V., Meyer, M., Pääbo, S., Popov, V., Režek, Ž., Sirakova, S., Skinner, M. W., Smith, G. M., Spasov, R., Talamo, S., Tuna, T., Wacker, L., Welker, F., Wilcke, A., Zahariev, N., McPherron, S. P., Tsanova, T., 2020. Initial Upper Paleolithic Homo Sapiens from Bacho Kiro Cave, Bulgaria. *Nature* 581, 299–302. https://doi.org/10.1038/s41586-020-2259-z
- Ichinose, N., Suga, E., Kadowaki, S., Tsukada, K., Nadmid, B., Umeda, K., Nuramkhaan, M., Massadeh, S., Henry, D. O., 2023. Petrographic and geochemical characterization of chert artifacts from Middle, Upper, and Epi-Paleolithic assemblages in the Jebel Qalkha area, southern Jordan. *Archaeometry* 65, 530–546. https://doi.org/10.1111/arcm.12824
- Inizan, M.-L., Reduron-Ballinger, M., Roche, H., Tixier, J., 1999. *Technology and terminology of knapped stone*. CREP, Nanderre.
- Jeyapoovan, T., Murugan, M., 2013. Surface roughness classification using image processing. *Measurement* 46, 2065–2072. https://doi.org/10.1016/j.measurement.2013.03.014
- Julig, P. J., Long, D. F. G., McFarland, V., Schroeder, B., 2007. Sourcing Of Archaeological Flints From The Middle To Upper Paleolithic Transition, At Jelf Al-Ajla Cave, Syria. In: Delage, C. (Ed.), *Chert Availability and Prehistoric Exploitation in the Near East.* John and Erica Hedges, Oxford, pp. 130–151.
- Kadowaki, S., 2013. Issues of Chronological and Geographical Distributions of Middle and Upper Palaeolithic Cultural Variability in the Levant and Implications for the Learning Behavior of Neanderthals and Homo sapiens. In: Akazawa, T., Nishiaki, Y., Aoki, K. (Eds.), *Dynamics of Learning in Neanderthals and Modern Humans Vol. 1: Cultural Perspectives*. Springer, New York. pp. 59–91. https://doi.org/10.1007/978-4-431-54511-8 4
- Kadowaki, S., Henry, D. O., 2019. Renewed Investigation of the Middle and Upper Paleolithic Sites in the Jebel Qalkha, Southern Jordan. In: Nakamura, S., Adachi, T., Abe, M. (Eds.), *Decades in Deserts: Essays on Near Eastern Archaeology in honour* of Sumio Fujii. Rokuichi Syobou, Tokyo, pp. 23–41.
- Kadowaki, S., Kurozumi, T., Henry, D. O., 2019a. Marine Shells from Tor Fawaz, Southern Jordan, and Their Implications for Behavioral Changes from the Middle to Upper Paleolithic in the Levant. In: Nishiaki, Y., Jöris, O. (Eds.), *Learning Among*

Neanderthals and Paleolithic Modern Humans: Archaeological Evidence. Springer, Singapore, pp. 161–178. https://doi.org/10.1007/978-981-13-8980-1 11

- Kadowaki, S., Omori, T., Nishiaki, Y., 2015. Variability in Early Ahmarian lithic technology and its implications for the model of a Levantine origin of the Protoaurignacian. *Journal of Human Evolution* 82, 67–87. https://doi.org/10.1016/j.jhevol.2015.02.017
- Kadowaki, S., Suga, E., Henry, D. O., 2021. Frequency and production technology of bladelets in Late Middle Paleolithic, Initial Upper Paleolithic, and Early Upper Paleolithic (Ahmarian) assemblages in Jebel Qalkha, Southern Jordan. *Quaternary International* 596, 4–21. https://doi.org/10.1016/j.quaint.2021.03.012
- Kadowaki, S., Tamura, T., Kida, R., Omori, T., Maher, L. A., Portillo, M., Hirose, M., Suga, E., Massadeh, S., Henry, D. O., 2022a. Lithic Technology and Chronology of Initial Upper Paleolithic Assemblages at Tor Fawaz, Southern Jordan. *Journal of Paleolithic Archaeology* 5, 1. https://doi.org/10.1007/s41982-021-00107-3
- Kadowaki, S., Tamura, T., Sano, K., Kurozumi, T., Maher, L. A., Wakano, J. Y., Omori, T., Kida, R., Hirose, M., Massadeh, S., Henry, D. O., 2019b. Lithic technology, chronology, and marine shells from Wadi Aghar, southern Jordan, and Initial Upper Paleolithic behaviors in the southern inland Levant. *Journal of Human Evolution* 135, 102646. https://doi.org/10.1016/j.jhevol.2019.102646
- Kadowaki, S., Tsukada, K., Hirose, M., Suga, E., Massadeh, S., Henry, D. O., 2022b. Survey for Chert Outcrops in the Western Hisma Basin, Southern Jordan and Its Implications for Paleolithic Raw Material Procurement. *ORIENT* 57, 3–20.
- Kalbe, J., Sharon, G., Porat, N., Zhang, C., Mischke, S., 2014. Geological setting and age of the Middle Paleolithic site of Nahal Mahanayeem Outlet (Upper Jordan Valley, Israel). *Quaternary International* 331, 139–148. https://doi.org/10.1016/j.quaint.2013.05.052
- Katz, O., Rechesa, Z., Roegiersc, J.-C., 2000. Evaluation of mechanical rock properties using a Schmidt Hammer. *International Journal of Rock Mechanics & Mining Sciences* 37, 723–728. https://doi.org/10.1016/S1365-1609(00)00004-6
- Kolobova, K. A., Kharevich, V. M., Kharevich, A. V., Fedorchenko, A. Y., Bocharova, E. N., Kurbanov, R., Flas, D., Krivoshapkin, A. I., Olsen, J. W., 2021. Archaeological and experimental studies of splintered pieces in the Central Asian Upper Paleolithic. *Archaeological and Anthropological Sciences* 13, 28. https://doi.org/10.1007/s12520-020-01256-2
- Kot, M., Pavlenok, K., Krajcarz, M. T., Pavlenok, G., Shnaider, S., Khudjanazarov, M., Leloch, M., Szymczak, K., 2020. Raw material procurement as a crucial factor determining knapping technology in the Katta Sai complex of Middle Paleolithic sites in the western Tian Shan piedmonts of Uzbekistan. *Quaternary International* 559, 97–109. https://doi.org/10.1016/j.quaint.2020.03.052
- Kot, M., Pavlenok, G., Krajcarz, M.T., Szymanek, M., Fedorowicz, S., Moska, P., Khudjanazarov, M., Szymczak, K., Leloch, M., Kogai, S., Talamo, S., Fewlass, H., Pavlenok, K., 2022. Is there Initial Upper Palaeolithic in Western Tian Shan? Example of an open-air site Katta Sai 2 (Uzbekistan). *Journal of Anthropological*

Archaeology 65, 101391. https://doi.org/10.1016/j.jaa.2021.101391

- Kuhn, S. L., 1995. *Mousterian Lithic Technology. An Ecological Perspective*. Princeton University Press, Princeton.
- Kuhn, S. L., 2004. Upper Paleolithic raw material economics at Üçağızlı cave, Turkey. *Journal of Anthropological Archaeology* 23, 431–448. https://doi.org/10.1016/j.jaa.2004.09.001
- Kuhn, S. L., 2019. Initial Upper Paleolithic: A (near) global problem and a global opportunity. *Archaeological Research in Asia* 17, 2–8. https://doi.org/10.1016/j.ara.2018.10.002
- Kuhn, S. L., 2020. *The Evolution of Paleolithic Technologies*. Routledge Taylor & Francis Group, London and New York. https://doi.org/10.4324/9781315642024
- Kuhn, S. L., Elston, R. G., 2002. Introduction: Thinking Small Globally. Archaeological Papers of the American Anthropological Association 12, 1–7. https://doi.org/10.1525/ap3a.2002.12.1.1
- Kuhn, S. L., Zwyns, N., 2014. Rethinking the initial Upper Paleolithic. *Quaternary International* 347, 29–38. https://doi.org/10.1016/j.quaint.2014.05.040
- Kuhn, S. L., Stiner, M. C., Güleç, E., Özer, I., Yılmaz, H., Baykara, I., Açıkkol, A., Goldberg, P., Molina, K. M., Ünay, E., Suata-Alpaslan, F., 2009. The early Upper Paleolithic occupations at Üçağızlı Cave (Hatay, Turkey). *Journal of Human Evolution* 56, 87–113. https://doi.org/10.1016/j.jhevol.2008.07.014
- Kuzmin, Y. V., Slavinsky, V. S., Tsybankov, A. A., Keates, S. G., 2022. Denisovans, Neanderthals, and Early Modern Humans: A Review of the Pleistocene Hominin Fossils from the Altai Mountains (Southern Siberia). *Journal of Archaeological Research* 30, 321–369. https://doi.org/10.1007/s10814-021-09164-2
- Lawn, B. R., 2021. Chipping: a pervasive presence in nature, science and technology. *Journal of Materials Science* 56, 8396–8405. https://doi.org/10.1007/s10853-020-05662-8
- Leder, D., 2016. Core reduction strategies at the initial upper paleolithic sites Ksar Akil and Abou Halka in Lebanon. *Lithics: The Journal of the Lithic Studies Society* 37, 33–54.
- Lerner, H. J., 2014a. Intra-raw material variability and use-wear formation: an experimental examination of a Fossiliferous chert (SJF) and a Silicified Wood (YSW) from NW New Mexico using the Clemex Vision processing frame. *Journal of Archaeological Science* 48, 34–45. https://doi.org/10.1016/j.jas.2013.10.030
- Lerner, H. J., 2014b. Intra-raw material variability and use-wear accrual: A continuing exploration. *Journal of Lithic Studies* 1, 165–186. https://doi.org/10.2218/jls.v1i1.755
- Lerner, H., Du, X., Costopoulos, A., Ostoja-Starzewski, M., 2007. Lithic raw material physical properties and use-wear accrual. *Journal of Archaeological Science* 34, 711–722. https://doi.org/10.1016/j.jas.2006.07.009
- Lewis, A. R., Williams, J. C., Buchanan, B., Walker, R. S., Eren, M. I., Bebber, M. R., 2022. Knapping quality of local versus exotic Upper Mercer chert (Ohio, USA) during the Holocene. *Geoarchaeology: An International Journal* 37, 486–496.

https://doi.org/10.1002/gea.21904

- Li, L., Lin, S. C., McPherron, S. P., Abdolahzadeh, A., Chan, A., Dogandžić, T., Iovita, R., Leader, G. M., Magnani, M., Rezek, Z., Dibble, H. L., 2022. A Synthesis of the Dibble et al. Controlled Experiments into the Mechanics of Lithic Production. *Journal of Archaeological Method and Theory*. https://doi.org/10.1007/s10816-022-09586-2
- Ligkovanlis, S., 2022. Hand-Preference and Lithic Production-Exploring Neanderthal Handedness Rates through the Study of Hertzian Fracture Features on Lithic Blanks. *Open Quaternary* 8, 4. https://doi.org/10.5334/oq.111
- Lin, S. C., Douglass, M. J., Holdaway, S. J., Floyd, B., 2010. The application of 3D laser scanning technology to the assessment of ordinal and mechanical cortex quantification in lithic analysis. *Journal of Archaeological Science* 37, 694–702. https://doi.org/10.1016/j.jas.2009.10.030
- Lin, S. C., Douglass, M. J., Mackay, A., 2016. Interpreting MIS3 Artifact Transport Patterns in Southern Africa Using Cortex Ratios: An Example from The Putslaagte Valley, Western Cape. South African Archaeological Bulletin 71, 173–180.
- Lin, S. C., McPherron, S. P., Dibble, H. L., 2015. Establishing statistical confidence in Cortex Ratios within and among lithic assemblages: a case study of the Middle Paleolithic of southwestern France. *Journal of Archaeological Science* 59, 89–109. https://doi.org/10.1016/j.jas.2015.04.004
- Luedtke, B. E., 1992. An Archaeologist's Guide to Chert and Flint. Institute of Archaeology, University of California, Los Angeles.
- Lyons, W. H., Glascock, M. D., Mehringer, Jr. P. J., 2003. Silica from sources to site: ultraviolet fluorescence and trace elements identify cherts from Lost Dune, southeastern Oregon, USA. *Journal of Archaeological Science* 30, 1139–1159. https://doi.org/10.1016/S0305-4403(03)00012-8
- Magnani, M., Rezek, Z., Lin, S. C., Chan, A., Dibble, H. L., 2014. Flake variation in relation to the application of force. *Journal of Archaeological Science* 46, 37–49. https://doi.org/10.1016/j.jas.2014.02.029
- Malinsky-Buller, A., Ekshtain, R., Hovers, E., 2014. Organization of lithic technology at 'Ein Qashish, a late Middle Paleolithic open-air site in Israel. *Quaternary International* 331, 234–247. https://doi.org/10.1016/j.quaint.2013.05.004
- Malinsky-Buller, A., Ekshtain, R., Munro, N., Hovers, E., 2021. Back to base: re-thinking variations in settlement and mobility behaviors in the Levantine Late Middle Paleolithic as seen from Shovakh Cave. Archaeological and Anthropological Sciences 13, 112. https://doi.org/10.1007/s12520-021-01313-4
- Marder, O., Goring-Morris, A. N., 2020. The Lithic Technologies of the Epipalaeolithic Hunter-Gatherers in the Negev, Israel: Implications from Refitting Studies. In: Leplongeon, A., Goder-Goldberger, M., Pleurdeau, D. (Eds.), Not Just a Corridor: Human occupation of the Nile Valley and neighbouring regions between 75,000 and 15,000 years ago. Muséum national d'Histoire naturelle, Paris, pp. 239–267.
- Marks, A. E., 1981. The Upper Paleolithic of the Levant. In: CNRS (Ed.), *Préhistoire du Levant*. Editions du Centre national de la recherche scientifique, Paris, pp. 369–372.

- Marks, A. E., 1983a. The Sites of Boker Tachtit and Boker: A Brief Introduction. Marks, A.E. (Ed.), *Prehistory and Paleoenvironments in the Central Negev, Israel: the Avdat/Aqev Area, Part 3.* Southern Methodist University, Dallas, pp. 15–37.
- Marks, A. E., 1983b. The Middle to Upper Paleolithic Transition in the Levant. *Advances in World Archaeology* 2, 51–98.
- Marks, A. E., Volkman, P. W., 1983. Changing Core Reduction Strategies: A Technological Shift from The Middle to The Upper Paleolithic in The Southern Levant. In: Trinkaus, E. (Ed.), *The Mousterian Legacy: Human Biological Change in the Upper Pleistocene*. B.A.R., Oxford, pp. 13–33.
- Marks, A. E., Shokler, J., Zilhão, J., 1991. Raw Material Usage in the Paleolithic: The Effects of Local Availability on Selection and Economy. In: Montet-White, A., Holen, S. (Eds.), *Raw Material Economies among Prehistoric Hunter-Gatherers*. University of Kansas, Lawrence KS, pp. 127–139.
- Matthews, J. A., Winkler, S., 2022. Schmidt-hammer exposure-age dating: a review of principles and practice. *Earth-Science Reviews* 230, 104038. https://doi.org/10.1016/j.earscirev.2022.104038
- McPherron, S. P., Braun, D. R., Dogandžić, T., Archer, W., Desta, D., Lin, S. C., 2014. An experimental assessment of the influences on edge damage to lithic artifacts: a consideration of edge angle, substrate grain size, raw material properties, and exposed face. *Journal of Archaeological Science* 49, 70–82. https://doi.org/10.1016/j.jas.2014.04.003
- McPherron, S. P., Abdolahzadeh, A., Archer, W., Chan, A., Djakovic, I., Dogandžić, T., Leader, G. M., Li, L., Lin, S., Magnani, M., Reeves, J., Rezek, Z., Weiss, M., 2020. Introducing platform surface interior angle (PSIA) and its role in flake formation, size and shape. *PLoS ONE* 15(11), e0241714. https://doi.org/10.1371/journal.pone.0241714
- Meignen, L., 2012. Levantine Perspectives on the Middle to Upper Paleolithic "Transition". *Archaeology, Ethnology and Anthropology of Eurasia* 40(3), 12–21. https://doi.org/10.1016/j.aeae.2012.11.003
- Meignen, L., 2019. The Mousterian Lithic Assemblages from Kebara Cave. In: Meignen, L., Bar-Yosef, O. (Eds.), Kebara Cave, Mt. Carmel, Israel. The Middle and Upper Paleolithic Archaeology. Part II. Peabody Museum Press, Cambridge, pp. 1–147.
- Miebach, A., Stolzenberger, S., Wacker, L., Hense, A., Litt, T., 2019. A new Dead Sea pollen record reveals the last glacial paleoenvironment of the southern Levant. *Quaternary Science Reviews* 214, 98–116. https://doi.org/10.1016/j.quascirev.2019.04.033
- Milne, S. B., Hamilton, A., Fayek, M., 2009. Combining Visual and Geochemical Analyses to Source Chert on Southern Baffin Island, Arctic Canada. *Geoarchaeology: An International Journal* 24, 429–449. https://doi.org/10.1002/gea.20273
- Moník, M., Hadraba, H., 2016. Mechanical characterization of raw material quality and its implication for Early Upper Palaeolithic Moravia. *Quaternary International* 425, 425–436. https://doi.org/10.1016/j.quaint.2016.08.042

- Moník, M., Hadraba, H., Milde, D., Chlup, Z., Nerudová, Z., Schnabl, P., 2021. Heat treatment and mechanics of Moravian Jurassic cherts. *Archaeological and Anthropological Sciences* 13, 158. https://doi.org/10.1007/s12520-021-01388-z
- Mraz, V., Fisch, M., Eren, M. I., Lovejoy, C. O., Buchanan, B., 2019. Thermal engineering of stone increased prehistoric toolmaking skill. *Scientific Reports* 9, 14591. https://doi.org/10.1038/s41598-019-51139-3
- Muller, A., Barkai, R., Shemer, M., Grosman, L., 2022. 3D morphology of handaxes from late Acheulean Jaljulia: a flexible reduction strategy in the Lower Paleolithic Levant. *Archaeological and Anthropological Sciences* 14, 206. https://doi.org/10.1007/s12520-022-01671-7
- Naito, Y. I., Hirose, M., Belmaker, M., Henry, D. O., Osawa, M., Nakazawa, T., Habinger, S. G., Tung, P., Bocherens, H., Massadeh, S., Kadowaki, S., 2022. Paleoenvironment and human hunting activity during MIS 2 in southern Jordan: Isotope records of prey remains and paleosols. *Quaternary Science Reviews* 282, 107432. https://doi.org/10.1016/j.quascirev.2022.107432
- Namen, A., Iovita, R., Nickel, K. G., Varis, A., Taimagambetov, Z., Schmidt, P., 2022. Mechanical properties of lithic raw materials from Kazakhstan: Comparing chert, shale, and porphyry. *PLoS ONE* 17, e0265640. https://doi.org/10.1371/journal.pone.0265640
- Neeley, M. P., 2007. Lithic Raw Material Availability And Use At The Epipaleplithic Site Of Tor Al-Tareeq In West-Central Jordan. In: Delage, C. (Ed.), *Chert Availability* and Prehistoric Exploitation in the Near East. John and Erica Hedges, Oxford, pp. 204–227.
- Negrino, F., Riel-Salvatore, J., 2018. From Neanderthals to Anatomically Modern Humans in Liguria (Italy): the current state of knowledge. In: Borgia, V., Cristiani, E. (Eds.), *Paleolithic Italy Advanced studies on early human adaptations in the Apennine peninsula*. Sidestone Press Academics, Leiden, pp. 161–182.
- Nelson, M. C., 1991. The Study of Technological Organization. Archaeological Method and Theory 3, 57–100.
- Nickel, K.G., Schmidt, P., 2022. Knapping force as a function of stone heat treatment. *PLoS ONE* 17(12), e0278643. https://doi.org/10.1371/journal.pone.0278643
- Nishiaki, Y., 2018. Initial Upper Paleolithic Elements of the Keoue Cave, Lebanon. In: Nishiaki, Y., Akazawa, T. (Eds.), *The Middle and Upper Paleolithic Archaeology of the Levant and Beyond*. Springer, Singapore, pp. 71–86. https://doi.org/10.1007/978-981-10-6826-3_6
- Nishiaki, Y., Kanjo, Y., Muhesen, S., Akazawa, T., 2012. The Temporal Variability of Late Levantine Mousterian Lithic Assemblages from Dederiyeh Cave, Syria. *Eurasian Prehistory* 9, 3–27.
- Odell, G. H., 2004. *Lithic analysis*. Springer, New York. https://doi.org/10.1007/978-1-4419-9009-9
- Ohnuma, K., Bergman, C. A., 1990. A Technological Analysis of the Upper Palaeolithic Levels (XXV-VI) of Ksar Akil, Lebanon. In: Mellars, P. (Ed.), *The Emergence of Modern Humans: An Archaeological Perspective*. Edinburgh University Press,

Edinburgh, pp. 91–138.

- Olszewski, D. I., 2006. Issues in the Levantine Epipaleolithic: The Madamaghan, Nebekian and Qalkhan (Levant Epipaleolithic). *Paléorient* 32, 19–26. https://doi.org/10.3406/paleo.2006.5168
- Pargeter, J., Shea, J. J., 2019. Going big versus going small: Lithic miniturization in hominin lithic technology. *Evolutionary Anthropology* 28, 72–85. https://doi.org/10.1002/evan.21775
- Parow-Souchon, H., Purschwitz, C., 2020. Variability in chert raw material procurement and use during the Upper Paleolithic and Early Neolithic of the southern Levant: A regional perspective from the Greater Petra area. *Journal of Archaeological Science: Reports* 29, 102087. https://doi.org/10.1016/j.jasrep.2019.102087
- Parow-Souchon, H., Hussain, S. T., Richter, J., 2021. Early Ahmarian Lithic Techno-Economy and Mobility at Al-Ansab 1, Wadi Sabra, Southern Jordan. *Journal of the Israel Prehistoric Society* 51, 6–64.
- Pastoors, A., Weniger, G.-C., Kegler, J. F., 2008. The Middle-Upper Paleolithic Transition at Yabroud II (Syria). A Re-Evaluation of The Lithic Material from The Rust Excavation. *Paléorient* 34, 47–65. https://doi.org/10.3406/paleo.2008.5256
- Phillips, J. L., 1988. The Upper Paleolithic of The Wadi Feiran, Southern Sinai. *Paléorient* 14, 183–200. https://doi.org/10.3406/paleo.1988.4467
- Rabb'a, I., 1987. Geological Map Sheet, 1:50000, Al Quwayra (3049 I). Hashemite Kingdom of Jordan, Ministry of Energy and Mineral Resources, National Resource Authority, Geology Directorate, Amman.
- Rabinovich, R., 2003. The Levantine Upper Paleolithic Faunal Record. In: Goring-Morris,
 A. N., Belfer-Cohen, A. (Eds.), *More Than Meets the Eye: Studies on Upper Paleolithic Diversity in the Near East*. Oxbow Books, Oxford, pp. 33–48.
- Radinović, M., Kajtez, I., 2021. Outlining the knapping techniques: Assessment of the shape and regularity of prismatic blades using elliptic Fourier analysis. *Journal of Archaeological Science: Reports* 38, 103079. https://doi.org/10.1016/j.jasrep.2021.103079
- Richter, J., Lift, T., Lehmkuhl, F., Hense, A., Hauck, T. C., Leder, D. F., Miebach, A., Parow-Souchon, H., Sauer, F., Schoenenberg, J., Al-Nahar, Hussain, S. T., 2020. Al-Ansab and the Dead Sea: Mid-MIS 3 archaeology and environment of the early Ahmarian population of the Levantine corridor. *PLoS ONE* 15, e0239968. https://doi.org/10.1371/journal.pone.0239968
- Richter, D., Schroeder, H. B., Pink, W. J., Julig, P. J., Schwarcz, H. P., 2001. The Middle to Upper Paleolithic Transition in the Levant and New Thermoluminescence Dates for A Late Mousterian Assemblage from Jerf al-Ajla Cave (Syria). *Paléorient* 27, 29–46. https://doi.org/10.3406/paleo.2001.4730
- Riel-Salvatore, J., Negrino, F., 2018. Human adaptations to climatic change in Liguria across the Middle-Upper Paleolithic transition. *Journal of Quaternary Science* 33, 313–322. https://doi.org/10.1002/jqs.3005
- Rodríguez-Rellán, C., 2016. Variability of the rebound hardness as a proxy for detecting the levels of continuity and isotropy in archaeological quartz. *Quaternary*

International 424, 191-211. https://doi.org/10.1016/j.quaint.2015.12.085

- Ronen, A., 2017. Tabun Cave in the Carmel Culture Sphere. In: Enzel, Y., Bar-Yosef, O. (Eds.), *Quarternary of the Levant: Environments, Climate Change, and Humans*. Cambridge University Press, New York, pp. 215–224. https://doi.org/10.1017/9781316106754.024
- Rose, J. I., Marks, A. E., 2014. "Out of Arabia" and the Middle-Upper Paleolithic transition in the southern Levant. *Quartär* 61, 49–85. https://doi.org/10.7485/QU61_03
- Sachpazis, C. I., 1990. Correlating schmidt hardness with compressive strength and young's modulus of carbonate rocks. *Bulletin of the International Association of Engineering Geology* 42, 75–83. https://doi.org/10.1007/BF02592622
- Sánchez de la Torre, M., Utrilla, P., Domingo, R., Jiménez, L., Le Bourdonnec, F.-X., Gratuze, B., 2020. Lithic raw material procurement at the Chaves cave (Huesca, Spain): A geochemical approach to defining Palaeolithic human mobility. *Geoarchaeology: An International Journal* 35, 856–870. https://doi.org/10.1002/gea.21808
- Schmidt, P., Hiscock, P., 2019. Evolution of Silcrete Heat Treatment in Australia—a Regional Pattern on the South-East Coast and Its Evolution over the Last 25 ka. *Journal of Paleolithic Archaeology* 2, 74–97. https://doi.org/10.1007/s41982-019-0020-7
- Schmidt, P., Morala, A., 2018. First Insights into the Technique Used for Heat Treatment of Chert at the Solutrean Site of Laugerie-Haute, France. *Archaeometry* 60, 885– 897. https://doi.org/10.1111/arcm.12358
- Schmidt, P., Buck, G., Berthold, C., Lauer, C., Nickel, K. G., 2019. The mechanical properties of heat-treated rocks: a comparison between chert and silcrete. *Archaeological and Anthropological Sciences* 11, 2489–2506. https://doi.org/10.1007/s12520-018-0710-y
- Schmidt, P., Masse, S., Laurent, G., Slodczyk, A., Le Bourhis, E., Perrenoud, C., Livage, J., Fröhlich, F., 2012. Crystallographic and structural transformations of sedimentary chalcedony in flint upon heat treatment. *Journal of Archaeological Science* 39, 135– 144. https://doi.org/10.1016/j.jas.2011.09.012
- Schmidt, P., Sanchez, O, S., Kind, C.-J., 2017. Stone heat treatment in the Early Mesolithic of southwestern Germany: Interpretation and identification. *PLoS ONE* 12(12), e0188576. https://doi.org/10.1371/journal.pone.0188576
- Schmidt, P., Stynder, D., Conard, N. J., Parkington, J. E., 2020. When was silcrete heat treatment invented in South Africa? *Palgrave Communications* 6, 73. https://doi.org/10.1057/s41599-020-0454-z
- Schyle, D., Richter, J. (Eds.), 2015. *Pleistocene Archaeology of the Petra Area in Jordan*. Verlag Marie Leidorf GmbH, Rahden/Westf.
- Shackley, M. S., 2008. Archaeological Petrology and The Archaeometry of Lithic Materials. Archaeometry 50, 194–215. https://doi.org/10.1111/j.1475-4754.2008.00390.x
- Sharon, G., 2008. The impact of raw material on Acheulian large flake production.

Journal of Archaeological Science 35, 1329–1344. https://doi.org/10.1016/j.jas.2007.09.004

Sharon, G., Oron, M., 2014. The lithic tool arsenal of a Mousterian hunter. *Quaternary International* 331, 167–185. http://dx.doi.org/10.1016/j.quaint.2013.10.024

Shea, J. J., 2008. Transitions or turnovers? Climatically-forced extinctions of Homo sapiens and Neanderthals in the east Mediterranean Levant. *Quaternary Science Reviews* 27, 2263–2270. https://doi.org/10.1016/j.quascirev.2008.08.015

- Shea, J. J., 2013. *Stone Tools in the Paleolithic and Neolithic Near East*. Cambridge University Press, New York. https://doi.org/10.1017/CBO9781139026314
- Shea, J. J., Sisk, M. L., 2010. Complex Projectile Technology and Homo sapiens Dispersal into Western Eurasia. *PaleoAnthropology* 2010, 100–122. https://doi.org/10.4207/PA.2010.ART36
- Shea, J. J., Stutz, A. J., Nilsson-Stutz, L., 2019. An Early Upper Paleolithic Stone Tool Assemblage from Mugar El-Hamamah, Jordan: An Interim Report. *Journal of Field Archaeology* 44(7), 1–17. https://doi.org/10.1080/00934690.2019.1655519
- Shemer, M., Boaretto, E., Greenbaum, N., Bar-Yosef Mayer, D. E., Tejero, J.-M., Langgut, D., Gnezdilov, D. L., Barzilai, O., Marder, O., Marom, N., 2023. Early Upper Paleolithic cultural variability in the Southern Levant: New evidence from Nahal Rahaf 2 Rockshelter, Judean Desert, Israel. *Journal of Human Evolution* 178, 103342. https://doi.org/10.1016/j.jhevol.2023.103342
- Sherwood, N. L., Li, H., Kuman, K., Li, C., 2018. Lithic raw material quality of Middle Pleistocene artefacts from the Han River, Danjiangkou Reservoir Region, central China. *Quaternary International* 480, 141–151. https://doi.org/10.1016/j.quaint.2017.01.009
- Shimelmitz, R., Kuhn, S. L., Weinstein-Evron, M., 2020. The evolution of raw material procurement strategies: A view from the deep sequence of Tabun Cave, Israel. *Journal of Human Evolution* 143, 102787. https://doi.org/10.1016/j.jhevol.2020.102787
- Shipton, C., 2023. Miniaturization and Abstraction in the Later Stone Age. *Biological Theory*. https://doi.org/10.1007/s13752-022-00423-z
- Shipton, C., Roberts, P., Archer, W., Armitage, S. J., Bita, C., Blinkhorn, J., Courtney-Mustaphi, C., Crowther, A., Curtis, R., d' Errico, F., Douka, K., Faulkner, P., Groucutt, H. S., Helm, R., Herries, A. I. R., Jembe, S., Kourampas, N., Lee-Thorp, J., Marchant, R., Mercader, J., Marti, A. P., Prendergast, M. E., Rowson, B., Tengeza, A., Boivin, N., 2018. 78,000-year-old record of Middle and Later Stone Age innovation in an East African tropical forest. *Nature Communications* 9, 1832. https://doi.org/10.1038/s41467-018-04057-3
- Shipton, C., Blinkhorn, J., Archer, W., Kourampas, N., Roberts, P., Prendergast, M. E., Curtis, R., Herries, A. I. R., Ndiema, E., Boivin, N., Petraglia, M. D., 2021. The Middle to Later Stone Age transition at Panga ya Saidi, in the tropical coastal forest of eastern Africa. *Journal of Human Evolution* 153, 102954. https://doi.org/10.1016/j.jhevol.2021.102954
- Simon, A. W., Coghlan, W. A., 1989. The Use of Indentation Testing to Obtain Precise

Hardness Measurements from Prehistoric Pottery. *American Antiquity* 54, 107–122. https://doi.org/10.2307/281334

- Slimak, L., 2023. The three waves: Rethinking the structure of the first Upper Paleolithic in Western Eurasia. *PLoS ONE* 18(5), e0277444. https://doi.org/10.1371/journal.pone.0277444
- Soto, M., Gómez de Soler, B., Vallverdú, J., 2018. The chert abundance ratio (CAR): a new parameter for interpreting Palaeolithic raw material procurement. *Archaeological and Anthropological Sciences* 10, 2027–2046. https://doi.org/10.1007/s12520-017-0516-3
- Speth, J. D., 1972. Mechanical Basis of Percussion Flaking. *American Antiquity* 37, 34–60. https://doi.org/10.2307/278884
- Spinapolice, E. E., 2012. Raw material economy in Salento (Apulia, Italy): new perspectives on Neanderthal mobility patterns. *Journal of Archaeological Science* 39, 680–689. https://doi.org/10.1016/j.jas.2011.10.033
- Stiner, M. C., Kuhn, S. L., Güleç, E., 2013. Early Upper Paleolithic shell beads at Üçağızlı Cave I (Turkey): Technology and the socioeconomic context of ornament lifehistories. *Journal of Human Evolution* 64, 380–398. https://doi.org/10.1016/j.jhevol.2013.01.008
- Stout, D., Quade, J., Semaw, S., Rogers, M. J., Levin, N. E., 2005. Raw material selectivity of the earliest stone toolmakers at Gona, Afar, Ethiopia. *Journal of Human Evolution* 48, 365–380. https://doi.org/10.1016/j.jhevol.2004.10.006
- Stutz, A. J., 2020. The Middle-Upper Paleolithic Transition: A Long-Term Biocultural Effect of Anatomically Modern Human Dispersal. In: Groucutt, H. S. (Ed.), *Cultural History and Convergent Evolution: Can We Detect Populations in Prehistory?* Springer, Cham, pp. 157–186. https://doi.org/10.1007/978-3-030-46126-3 9
- Suga, E., Ichinose, N., Tsukada, K., Kadowaki, S., Massadeh, S., Henry, D.O., 2022. Investigating changes in lithic raw material use from the Middle Paleolithic to the Upper Paleolithic in Jebel Qalkha, southern Jordan. *Archaeological Research in Asia* 29, 100347. https://doi.org/10.1016/j.ara.2021.100347
- Suga, E., Tsukada, K., Tarawneh, O., Massadeh, S., Kadowaki, S., 2023. Explaining the increase in "high-quality chert" in the Early Upper Paleolithic artifacts in southern Jordan: Quantitative examination of chert mechanical properties and fracture predictability. *Journal of Paleolithic Archaeology* 6, 35. https://doi.org/10.1007/s41982-023-00164-w
- Tarawneh, K., 2002. Geological Map Sheet, 1:50,000, Ma'an (3150III). Hashemite Kingdom of Jordan, Ministry of Energy and Mineral Resources, National Resource Authority, Geology Directorate, Amman.
- Tarawneh, K., 2019. The Geology, Stratigraphy and Structure of Ma`an Area, South Jordan. *Journal of Environment and Earth Science* 9, 103–117.
- Terradillos-Bernal, M., Rodríguez-Álvarez, X., 2017. The influence of raw material quality on the characteristics of the lithic tool edges from the Atapuerca sites (Burgos, Spain). *Quaternary International* 433, 211–223. https://doi.org/10.1016/j.quaint.2015.10.122

- Teymen, A., 2021. Statistical models for estimating the uniaxial compressive strength and elastic modulus of rocks from different hardness test methods. *Heliyon* 7, e06891. https://doi.org/10.1016/j.heliyon.2021.e06891
- Thiry, M., Fernandes, P., Milnes, A., Raynal, J., 2014. Driving forces for the weathering and alteration of silica in the regolith: Implication for studies of prehistoric flint tools. *Earth-Science Reviews* 136, 141–154. https://doi.org/10.1016/j.earscirev.2014.05.008
- Tomasso, A., Porraz, G., 2016. Hunter-Gatherer Mobility and Embedded Raw-Material Procurement Strategies in the Mediterranean Upper Paleolithic. *Evolutionary Anthropology* 25, 164–174. https://doi.org/10.1002/evan.21488
- Tomasso, A., Rots, V., 2021. Looking into Upper Paleolithic gear: The potential of an integrated techno-economic approach. *Journal of Anthropological Archaeology* 61, 101240. https://doi.org/10.1016/j.jaa.2020.101240
- Torfstein, A., Goldstein, S. L., Kushnir, Y., Enzel, Y., Haug, G., Stein, M., 2015. Dead sea drawdown and monsoonal impacts in the Levant during the last interglacial. *Earth* and *Planetary Science Letters* 412, 235–244. https://doi.org/10.1016/j.epsl.2014.12.013
- Tostevin, G. B., 2012. Seeing Lithics. Oxbow Books, Oxford and Oakville.
- Tsirk, A., 2014. Fracture in Knapping. Archaeopress, Oxford.
- Tsobgou, A.R., Dabard, M.-P., 2010. Petrographical, Structural, and Mechanical Analysis of Armorican Phtanites: A Key Raw Material for the Mesolithic in Western France. *Geoarchaeology: An International Journal* 25, 327–351. https://doi.org/10.1002/gea.20310
- Tsukada, K., 2018. Quantitative analysis of chemical composition of anhydrous silicate minerals using SEM–EDX installed at Nagoya University Museum. *Bulletin of the Nagoya University Museum* 34, 1–10. https://doi.org/10.18999/bulnum.034.01
- Valde-Nowak P., Cieśla M., 2020. Models of Raw Material Exploitation as an Indicator of Middle Paleolithic Mobility: Case Studies from Uplands of Northern Central Europe. In: Cascalheira J., Picin A. (Eds.), *Short-Term Occupations in Paleolithic Archaeology*. Springer, Cham, pp. 105–120. https://doi.org/10.1007/978-3-030-27403-0 5
- Valletta, F., Smilansky, U., Goring-Morris, A. N., Grosman, L., 2020. On measuring the mean edge angle of lithic tools based on 3-D models - a case study from the southern Levantine Epipaleolithic. *Archaeological and Anthropological Sciences* 12, 49. https://doi.org/10.1007/s12520-019-00954-w
- Vallini, L., Marciani, G., Aneli, S., Bortolini, E., Benazzi, S., Pievani, T., Pagani, L., 2022. Genetics and Material Culture Support Repeated Expansions into Paleolithic Eurasia from a Population Hub Out of Africa. *Genome Biology and Evolution* 14, evac045. https://doi.org/10.1093/gbe/evac045
- Varoner, O., Marder, O., Orbach, M., Yeshurun, R., Zaidner, Y., 2022. Lithic provisioning strategies at the Middle Paleolithic open-air site of Nesher Ramla, Israel: A case study from the upper sequence. *Quaternary International* 624, 19–33. https://doi.org/10.1016/j.quaint.2021.08.008

- Vujević, D., Perhoč, Z., Ivančić, T., 2017. Micro-Mousterian in Northern Dalmatia. *Quaternary International* 450, 50–67. http://dx.doi.org/10.1016/j.quaint.2016.11.019
- Webb, J. A., Domanski, M., 2008. The relationship between lithology, flaking properties and artefact manufacture for Australian silcretes. *Archaeometry* 50, 555–575. https://doi.org/10.1111/j.1475-4754.2007.00381.x
- Whitehouse, D. J., 1997. Surface metrology. *Measurement Science and Technology* 8, 955–972. https://dx.doi.org/10.1088/0957-0233/8/9/002
- Whittaker, J. C., 1994. *Flintknapping: Making and Understanding Stone Tools*. University of Texas Press, Austin.
- Williams, J. C., Simone, D. M., Buchanan, B., Boulanger, M. T., Bebber, M. R., Eren, M. I., 2019. Nine-thousand years of optimal toolstone selection through the North American Holocene. *Antiquity* 93(368), 313–324. https://doi.org/10.15184/aqy.2018.187
- Wilson, L., 2007. Understanding Prehistoric Lithic Raw Material Selection: Application of a Gravity Model. *Journal of Archaeological Method and Theory* 14, 388–411. https://doi.org/10.1007/s10816-007-9042-4
- Wilson, L., Agam, A., Barkai, R., Gopher, A., 2016. Raw material choices in Amudian versus Yabrudian lithic assemblages at Qesem Cave: A preliminary evaluation. *Quaternary International* 398, 61–69. https://doi.org/10.1016/j.quaint.2015.02.015
- Winkler, E. M., 1975. Stone: Properties, Durability in Man's Environment. Springer, New York. https://doi.org/10.1007/978-3-7091-4120-5
- Winkler, S., Matthews, J. A., 2014. Comparison of electronic and mechanical Schmidt hammers in the context of exposure-age dating: are Q- and R-values interconvertible? *Earth Surface Processes and Landforms* 39, 1128–1136. https://doi.org/10.1002/esp.3584
- Yagiz, S., 2009. Predicting uniaxial compressive strength, modulus of elasticity and index properties of rocks using the Schmidt hammer. *Bulletin of Engineering Geology and the Environment* 68, 55–63. https://doi.org/10.1007/s10064-008-0172-z
- Yasar, E., Erdogan, Y., 2004. Correlating sound velocity with the density, compressive strength and Young's modulus of carbonate rocks. *International Journal of Rock Mechanics & Mining Sciences* 41, 871–875. https://doi.org/10.1016/j.ijrmms.2004.01.012
- Yonekura, K., 2010. Experimental Study on Heat Alteration of Palaeolithic Material: Preliminary Results from Shale in the Northeastern Region of Japan. Asian Perspectives 49, 348–362. https://doi.org/10.1353/asi.2010.0011
- Yonekura, K., Suzuki, T., 2009. Microhardness analysis and characterization of Palaeolithic stone tool materials for understanding primary material selections and utilizations. *Materials Characterization* 60, 282–291. https://doi.org/10.1016/j.matchar.2008.08.008
- Zhang, Z. X., 2002. An empirical relation between mode I fracture toughness and the tensile strength of rock. *International Journal of Rock Mechanics & Mining Sciences* 39, 401–406. https://doi.org/10.1016/S1365-1609(02)00032-1

- Zhou, Z., Huan, Y., Shao, Y., Dai, Y., Yang, H., 2014. Heat treated stone artifacts at Shuidonggou, Northwest China and their archaeological implications. *Quaternary International* 347, 84–90. https://doi.org/10.1016/j.quaint.2014.07.010
- Zwyns, N., 2021. The Initial Upper Paleolithic in Central and East Asia: Blade Technology, Cultural Transmission, and Implication for Human Dispersals. *Journal* of Paleolithic Archaeology 4, 19. https://doi.org/10.1007/s41982-021-00085-6

Appendix



Fig. S1. Schmidt Hammer mechanism. (a) The instrument is ready for test. (b) A body pushed toward a rock sample and a spring is stretched. (c) A steel hammer is released and the rock sample is loaded. (d) The steel hammer rebounds.



Fig. S2. Rockwell Hardness (HRC) mechanism. (a) The surface of rock sample is pushed by a diamond conical indenter with reference load (98.07 N). (b) The indenter pushed with test load (1471 N). (c) The difference in groove depth caused by the two loads is measured.

Туре	M1	M2	M3	M4	M5
Texture (R_a value)	Medium (≥2)				
Translucence	Medium-Low	Medium-Low	Medium-Low	Medium-Low	Low
Color	Light Gray	Light Gray	Light Gray	Light Gray	Black
	L. Brn. Gray	L. Brn. Gray	L. Brn. Gray	L. Brn. Gray	V. Dk. Gray
	Grayish Brn.	Grayish Brn.	Grayish Brn.	Grayish Brn.	
	L. Ye. Orange	L. Ye. Orange	L. Ye. Orange	L. Ye. Orange	
	Dull Orange	Dull Orange	Dull Orange	Dull Orange	
	Dull Brown	Dull Brown	Dull Brown	Dull Brown	
	Brown	Brown	Brown	Brown	
Munsell Notation	7.5 YR 8/2-4/2	7.5 YR 8/2-4/2	7.5 YR 8/2-4/2	7.5 YR 8/2-4/2	10YR 2/1-3/1
	7.5 YR 8/3-4/3	7.5 YR 8/3-4/3	7.5 YR 8/3-4/3	7.5 YR 8/3-4/3	
UV signature	Orange 590	Yellow 570	Dk. Violet 400	Yellow 550	Dk. Violet 400
Microfossils	Much-Few	Much-Few	Much-Few	Much-Few	Much-Few
SiO ₂ content	High-Medium	High-Medium	High-Medium	High-Medium	High-Medium
CaO content	Little	Little	Little	Little	Little
Chert Variety No. by	1 or 3	2 or 4	NA	NA	5
Henry and Mraz, 2020					

Table S1. Macroscopic, microscopic, and chemical attributes of chert types defined in this study for the lithic assemblages from the Jebel Qalkha area.

Table S1. (Continued)

Туре	FL1	FL2	FL3	FH1	FH2	NA
Texture (R_a value)	Fine (below 2)	Heavy burned				
Translucence	Medium-Low	Medium-Low	Low	High	High	or weathered
Color	Light Gray	Brn. Gray	Black	Dk. Dusky	Red Gray	Quarzite
	Grayish Brn.	Grayish Ye.	V. Dk. Gray	Red	Weak Red	Other Catchall
		Brwon				
Munsell Notation	10 YR 5/2-7/2	10 YR 5/1-4/1	10YR 2/1-3/1	5 YR 3/3-5/3	5-10 YR 4-5/2	
		10 YR 5/2-4/2				
UV signature	Indigo 450	Indigo 450	Dk. Violet 400	Indigo 450	Orange 590	
Microfossils	Few-None	Few-None	Few-None	None	None	
SiO ₂ content	High–Medium	High-Medium	High–Medium	High	High	
CaO content	Little–None	Little–None	Little–None	None	None	
Chert Variety No. by	7	NA	NA	6	9	10 or 11
Henry and Mraz, 2020						

Table S2. Measurement values of rebound hardness. Some samples are not included in the list because they broke during the measurement and could not be measured 10 times.

No O	Outorona	Davy Matarial type	Mass (kg)	Measurement values											
INO.	Outerops	Kaw Material type		1st	2nd	3rd	4th	5th	6th	7th	8th	9th	10th		
1	Abbasiya	Type M chert	21.4	71	81	75	80	70	82	81	71	78	82		
2	Abbasiya	Type FH chert	61.4	84	83	78	80	80	75	73	82	83	80		
3	Abbasiya	Type M chert	20.0	75	78	72	79	80	81	76	80	77	78		
4	Abbasiya	Type FH chert	20.0	85	82	83	80	79	78	74	78	77	80		
5	Abbasiya	Type M chert	9.2	68	69	78	73	78	75	76	70	72	75		
7	Abbasiya	Type M chert	24.4	84	84	82	81	81	75	81	82	82	82		
8	Abbasiya	Type FH chert	23.4	80	80	81	81	79	81	74	80	81	79		
9	Abbasiya	Type M chert	17.2	73	71	71	70	69	70	73	70	72	78		
10	Abbasiya	Type M chert	24.4	72	82	75	78	81	79	74	74	77	83		
11	Abbasiya	Type FH chert	24.6	85	85	76	75	72	80	78	75	76	81		
13	Abbasiya	Type FH chert	13.6	72	80	73	73	79	75	71	77	74	73		
14	Abbasiya	Type FH chert	12.2	79	73	72	69	75	69	76	77	76	79		
15	Abbasiya	Type FH chert	31.8	86	85	84	77	77	82	83	77	80	82		
16	Abbasiya	Type M chert	18.6	80	73	82	78	77	81	74	78	73	77		
17	Abbasiya	Type FH chert	21.2	76	75	77	78	81	75	78	76	80	81		
18	Wadi Abu Sawwan	Type M chert	18.2	70	75	76	70	79	81	75	76	73	81		
19	Wadi Abu Sawwan	Type M chert	41.8	82	78	76	81	82	79	74	75	78	81		
21	Wadi Abu Sawwan	Type M chert	42.6	86	84	83	77	78	76	80	80	79	80		
22	Wadi Abu Sawwan	Type M chert	13.4	73	74	79	75	72	78	77	78	79	75		
23	Wadi Abu Sawwan	Type M chert	36.2	78	79	82	83	75	78	80	79	74	77		
26	Humayma	Type FH chert	35.0	86	84	79	81	77	77	76	80	77	80		

No. October			Measurement values										
No.	Outcrops	Raw Material type	Mass (kg)	1st	2nd	3rd	4th	5th	6th	7th	8th	9th	10th
29	Ma'an Plateau	Type FH chert	14.2	76	82	78	79	82	75	77	75	78	82
30	Abbasiya	Type FH chert	5.2	84	82	75	76	76	80	76	78	83	76
31	Wadi Abu Sawwan	Type FH chert	4.8	79	79	79	80	77	81	75	81	75	80
32	Abbasiya	Type FH chert	4.6	71	74	79	74	74	73	79	73	74	71
34	Humayma	Type M chert	5.6	78	77	77	75	71	77	73	74	75	75
35	Ma'an Plateau	Type M chert	4.8	73	72	75	71	73	71	72	75	71	72
40	Hokkaido	Obsidian	8.6	84	83	82	79	77	75	75	74	68	67
41	Hokkaido	Obsidian	18.3	82	78	78	77	77	76	76	74	71	71
42	Hokkaido	Obsidian	6.3	81	80	78	76	74	73	65	61	54	45
43	Hokkaido	Obsidian	12.0	84	83	82	81	80	79	71	69	69	61
44	Hokkaido	Obsidian	10.3	83	81	80	78	78	67	67	66	61	59
45	Hokkaido	Obsidian	5.0	79	79	79	78	77	77	74	73	73	70
46	Hokkaido	Obsidian	20.0	84	82	81	81	80	79	76	76	66	64
47	Hokkaido	Obsidian	18.0	87	86	85	85	84	83	82	81	78	66
48	Hokkaido	Obsidian	9.0	86	80	80	80	80	77	71	68	60	47
49	Hokkaido	Andesite	19.9	63	63	59	59	57	56	54	53	51	47
50	Hokkaido	Andesite	17.0	71	70	70	69	69	68	68	68	68	66
51	Hokkaido	Andesite	15.7	66	66	65	64	63	63	62	62	59	59
52	Hokkaido	Andesite	11.5	65	62	61	61	61	58	58	54	50	51
53	Hokkaido	Andesite	6.0	71	71	71	69	69	67	67	66	64	57
54	Hokkaido	Andesite	12.9	57	54	53	52	50	49	46	45	44	42
55	Hokkaido	Andesite	14.7	72	72	71	71	69	68	67	64	61	56

Table S2. (Continued)

94

NI-	Raw Material	Mother	SiO ₂	LOI	Outcrop	Measurement value										
INO.	Туре	rock ID	(wt.%)	(wt.%)		1st	2nd	3rd	4th	5th	6th	7th	8th	9th	10th	
1	Type FH chert	А	97.34	1.40	Ma'an Plateau	38.7	36.5	44.2	26.4	46.1	48.9	30.7	37.3	19.9	-8.0	
2	Type FH chert	А	97.34	1.40	Ma'an Plateau	-11.0	33.3	43.1	14.8	39.7	5.9	-3.2	44.7	53.2	1.5	
3	Type FH chert	В	96.67	1.21	Ma'an Plateau	39.6	15.1	42.1	9.6	-10.2	42.2	11.9	35.7	27.5	33.7	
4	Type FH chert	В	96.67	1.21	Ma'an Plateau	23.7	16.6	45.9	31.7	17.7	7.2	18.1	16.2	30.8	20.0	
5	Type FH chert	В	96.67	1.21	Ma'an Plateau	44.5	24.3	1.9	43.0	40.2	21.9	44.8	31.8	28.3	48.5	
6	Type FH chert	С	97.41	0.80	Humayma	15.8	46.5	40.4	31.9	22.8	25.6	34.1	-5.6	52.0	20.0	
7	Type FH chert	С	97.41	0.80	Humayma	44.3	52.1	43.5	31.6	53.0	52.0	46.5	62.7	53.8	-6.8	
8	Type FH chert	D	97.39	1.13	Abbasiya	32.9	31.1	-16.9	-2.5	33.7	33.2	-38.2	33.3	-29.4	26.0	
9	Type FH chert	D	97.39	1.13	Abbasiya	22.7	38.7	35.8	34.7	13.6	64.2	39.9	29.7	3.4	39.3	
10	Type FH chert	D	97.39	1.13	Abbasiya	39.0	-10.4	43.9	11.2	21.9	36.7	9.3	27.7	52.5	-2.8	
11	Type M chert	E	94.93	1.63	Humayma	57.5	29.1	46.0	57.8	62.4	50.2	29.2	51.7	42.8	45.4	
12	Type M chert	E	94.93	1.63	Humayma	26.3	31.6	36.4	23.3	39.4	29.1	46.5	37.0	56.0	42.9	
13	Type M chert	E	94.93	1.63	Humayma	1.6	41.0	25.4	26.2	33.0	23.4	34.5	11.6	47.6	15.7	
14	Type M chert	F	94.74	1.54	Humayma	67.1	-2.4	55.5	64.3	53.2	59.2	65.8	34.6	58.6	52.4	
15	Type M chert	F	94.74	1.54	Humayma	59.2	12.6	22.4	33.8	51.8	57.3	43.1	51.7	63.1	53.2	
16	Type M chert	F	94.74	1.54	Humayma	50.9	46.7	35.8	30.2	54.1	51.8	10.6	36.1	30.8	35.1	
17	Obsidian	—	—	—	Hokkaido	-45.3	-10.4	34.0	2.8	-7.8	31.6	-2.4	36.9	23.4	1.9	
18	Obsidian	—	—	—	Hokkaido	3.6	-40.5	12.9	-2.3	8.7	37.1	43.9	-49.2	28.0	15.0	
19	Obsidian	—	—	—	Hokkaido	25.8	4.8	-1.0	-7.8	-10.7	28.7	18.8	22.4	0.0	34.0	
20	Glass block	—	—		—	-42.6	-16.1	-11.6	-25.6	-51.6	-24.7	-37.6	-10.9	-51.3	-18.8	

Table S3. Measurement values of Rockwell Hardness (HRC), and SiO₂ and loss on ignition (LOI) percentage.